

CEP Discussion Paper No 1544

May 2018

Weather, Labor Reallocation and Industrial Production: Evidence from India

Jonathan Colmer





Abstract

Temperature-driven reductions in the demand for agricultural labor are associated with increases in the share of workers engaged in manufacturing, suggesting that the ability of non-agricultural sectors to absorb workers may play a key role in attenuating the economic consequences of weather-driven changes in agricultural productivity. Exploiting firm-level variation in the propensity to absorb these workers, I find that this reallocation is associated with relative expansions in manufacturing activity in exible labor market environments. Counter-factual estimates suggest that in the absence of labor reallocation the aggregate consequences of temperature increases would be up to 40% higher.

Key words: labor reallocation, agricultural productivity, labor regulation, industrial production. JEL: O13; Q54; J62

This paper was produced as part of the Centre's Urban and Spatial Programme. The Centre for Economic Performance is financed by the Economic and Social Research Council.

Temperature-driven reductions in the demand for agricultural labor are associated with increases in the share of workers engaged in manufacturing, suggesting that the ability of non-agricultural sectors to absorb workers may play a key role in attenuating the economic consequences of weather-driven changes in agricultural productivity. Exploiting firm-level variation in the propensity to absorb these workers, I find that this reallocation is associated with relative expansions in manufacturing activity in exible labor market environments. Counter-factual estimates suggest that in the absence of labor reallocation the aggregate consequences of temperature increases would be up to 40% higher.

Jonathan Colmer, Department of Economics, University of Virginia and Centre for Economic Performance, London School of Economics.

Published by Centre for Economic Performance London School of Economics and Political Science Houghton Street London WC2A 2AE

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means without the prior permission in writing of the publisher nor be issued to the public or circulated in any form other than that in which it is published.

Requests for permission to reproduce any article or part of the Working Paper should be sent to the editor at the above address.

© J. Colmer, submitted 2018.

1 Introduction

Understanding the relationship between economic and natural systems is of central importance, especially in developing countries given the role that agriculture plays for the economic lives of the poor. One relationship that has received particular attention in recent years is the relationship between weather and economic activity. However, less is known about how economic agents respond to weather, and the degree to which behavioral responses moderate the economic consequences of weather. One important margin is the degree to which agricultural workers are able to manage weather-driven changes in labor demand. Are workers able to find work in other sectors or locations, or do labor market frictions impede this reallocation, inextricably coupling the livelihoods and welfare of these workers to changes in the natural environment?

I seek to answer this question by combining worker-, firm- and district-level data with high-resolution meteorological data in India, exploring the effects of weather on agricultural productivity, industrial production and local labor market outcomes. First, and unsurprisingly, I identify that increases in temperature are associated with a reduction in agricultural production, and in turn the employment and wages of agricultural workers, demonstrating the important role that weather plays in driving short-run agricultural productivity, and the livelihoods of agricultural workers (Deaton, 1992, Paxson, 1992, Rosenzweig and Binswanger, 1993, Townsend, 1994, Jayachandran, 2006, Guiteras, 2009, Taraz, 2012, Kaur, 2014, Mobarak and Rosenzweig, 2014, Kala, 2015).

Having observed the negative effects of temperature increases on agricultural outcomes, it is important to understand what happens to workers in response to these changes. While I find that weather is a strong driver of short-run agricultural productivity, I observe that it has no effect on agricultural prices, consistent with a "law of one price", suggesting that reductions in agricultural productivity should push workers out of agriculture and into other tradable sectors of the economy (Burgess and Donaldson, 2010, 2012, Donaldson, 2015b); however, this depends on the ability of workers to move across sectors, and on the ability of other sectors to absorb these workers in response to short-run productivity shocks. Having shown that increases in temperature result in a reduction in the employment share of agriculture, I estimate a corresponding shift of labor into the manufacturing sector. In addition, I estimate that there are no changes in unemployment, or in the local population through migration, bounding local labor markets and suggesting that the main margin through which labor reallocation in India occurs is sectoral rather than spatial. These results suggest that the ability of non-agricultural sectors to absorb workers within local labor markets may play a key role in managing the economic consequences of weather-driven changes in agricultural productivity, highlighting the importance of market integration and diversification can play in attenuating sectoral productivity shocks (Matsuyama, 1992, Foster and Rosenzweig, 2004, Jayachandran, 2006, Burgess and Donaldson, 2010, 2012, Autor et al., 2013, Bustos et al., 2015, Costinot et al., 2015, Donaldson, 2015a,b, Henderson et al., 2015, Hornbeck and Keskin, 2015, Hornbeck and Moretti, 2015, Mian and Sufi, 2015).

Having demonstrated that other sectors of the economy are major absorbers of labor in the face of weather-driven changes in agricultural productivity, it is of interest to understand what these workers do, and how they affect firm and incumbent worker outcomes when they move into the manufacturing sector. However, identifying these effects presents a number of empirical challenges. To interpret the effects of weather on manufacturing outcomes as being driven by labor reallocation, it is necessary that these outcomes are not affected by weather in any other way. This is a strong assumption, as there are potentially many channels through which weather could affect other sectors, directly and/or through agricultural linkages.¹ Consequently, any estimate of the relationship between weather and manufacturing outcomes will provide the net effect of all the competing and complementary channels involved. Given this ambiguity, it is difficult to interpret empirical estimates of weather in a meaningful way. Where empirically relevant channels move in the same direction, we fail to arrive at a meaningful economic interpretation. Where multiple channels are competing, specific effects may be missed entirely, or selected interpretations underestimated.

To try and address this concern, I exploit variation in the propensity of firms to absorb labor in response to transitory changes in labor demand, arising from year-to-year changes in the weather, helping to identify the channel of interest – the labor reallocation effect. To do this, I construct a firm-level measure of exposure to India's labor regulation environment that builds on Besley and Burgess (2004), who classify the rigidity of the labor

¹Changes in agricultural productivity could affect manufacturing outcomes in sectors that use agricultural products as inputs, propagating shocks through intermediaries (Acemoglu et al., 2012), and a reduction in agricultural income could reduce the consumption base for manufactured products with local demand (Soderbom and Rijkers, 2013, Henderson et al., 2012, Santangelo, 2015, Emerick, 2016). Weather may also affect manufacturing production directly through its impact on factors of production. For example, an increase in temperature may reduce production through a reduction in the health or physical/cognitive ability of workers and managers, through an increase in absenteeism due to avoidance behaviour (Mackworth, 1946, 1947, Kenrick and McFarlane, 1986, Hsiang, 2010, Cachon et al., 2012, Adhvaryu et al., 2015, Burgess et al., 2014, Somonathan et al., 2015, Heal and Park, 2014, Graff Zivin and Neidell, 2014, Graff Zivin et al., 2015). Heavy rainfall may affect workers' ability to get to work (Bandiera et al., 2015b), disrupt supply chains. In addition, increased temperature, or a reduction in rainfall in areas dependent on hydroelectric power generation, is likely to put additional stress on an already fragile electricity infrastructure, reducing the supply of electrical power (Ryan, 2014, Alcott et al., 2015). Increases in temperature or reductions in rainfall may increase groundwater use, resulting in competition for water between agriculture and industry (Keskin, 2010). Finally, capital stocks and flows may be affected if weather affects capital depreciation, the relative productivity of inputs, or the level of investment in the economy if capital is locally constrained (Jina and Hsiang, 2015, Asher and Novosad, 2014).

market environment using state-level amendments to the Industrial Disputes Act of 1947 (hereafter IDA). In rigid labor market environments, firms face significant hiring and firing costs that, I argue, diminish the incentive to hire workers in response to transitory changes in the availability of labor (Oi, 1962, Nickell, 1978, Bentolila and Bertola, 1990, Hamermesh, 1993, Heckman, 2003, Besley and Burgess, 2004, Haltiwanger et al., 2008, Ahsan and Pagés, 2009, Adhvaryu et al., 2013, Amirapu and Gechter, 2014, Chaurey, 2015). By contrast, these costs are supposedly lower in more flexible labor market environments, where firms have more bargaining power over hiring decisions. However, this alone is not sufficient to identify the effects of labor reallocation. There may be other differences across space that could conflate the estimated effects of weather on manufacturing outcomes based on these differences. Consequently, I introduce firm-level exposure to the labor regulation environment, based on chapter 5b of the Industrial Disputes Act, which specifies the size that firms can reach before they are regulated under the IDA. In support of this identification strategy, I observe bunching in the firm-size distribution to the left of the regulatory threshold in rigid labor market environments, but not in the case of the flexible labor market environments, suggesting that the IDA has a binding effect on firm behavior.

For the identification strategy to have any viability the other effects of weather must not have a differential effect across across labor regulation environments. To test this I first examine the effects of temperature on unregulated firms. This provides a direct test for the assumption that the other channels of weather are not differential across labor regulation environments. In support of this assumption I estimate that there are limited differential effects of temperature on unregulated firms across labor regulation environments. While it would be a overtly strong to claim the absence of confounding factors, this suggests that, at a minimum, the net effect of other policy variation, heterogeneous weather effects and general equilibrium considerations cancel each other out across labor regulation environments. Further support for this premise is found by exploring the differential effects of temperature on unregulated sectors across labor regulation environments, where again I find limited evidence of spatial differences. Consequently, to the extent that these factors do exist, empirically they are likely to have a limited impact on the identification of the labor reallocation effect in regulated firms. Only confounding differences across labor regulation environments that differentially affect regulated firms, but not unregulated firms, will affect identification of the labor reallocation effect.

With this in mind, I next examine the effects of temperature on regulated firms. I estimate that, in rigid labor market environments, an increase in temperature is associated with a negative impact on firm performance, consistent with – but not limited to – an emerging literature that suggests that increases in temperature have significant effects on labor productivity through a drag on physiological and cognitive ability (Mackworth, 1946, 1947, Kenrick and McFarlane, 1986, Hsiang, 2010, Cachon et al., 2012, Adhvaryu et al., 2015, Somonathan et al., 2015, Heal and Park, 2014, Graff Zivin and Neidell, 2014, Graff Zivin et al., 2015). However, in flexible labor market environments, I estimate that firms experience a relative increase in employment and output, with new entrants moving into casual manufacturing activities. This expansion offsets the adverse effects of temperature. These effects provide support for the premise that firms in flexible labor market environments are more able to absorb workers in response to agricultural productivity shocks. In addition I estimate a relative increase in the average wage of permanent workers, manufacturing productivity (TFP and output per worker), and the number of items that the firm produces, suggesting that the activities that casual and permanent workers engage in are complementary to production.

The absence of movement into permanent positions suggests that labor markets can be characterised, at least in the short run, as dualistic: workers earn different wages depending on the type of employment activities in which they engage (casual vs. permanent).² These results are consistent with an emerging literature that explores the impact of agricultural productivity shocks on local economic activity (Hornbeck, 2012, Hornbeck and Naidu, 2014, Bustos et al., 2015, Henderson et al., 2015, Hornbeck and Keskin, 2015, Marden, 2015). However, most of the research to date has focused on long-run changes in agricultural productivity due to permanent changes in technology or the environment. By focusing on short-run changes in the weather, it is plausible that other factors of production, such as capital or the allocation of land, are held constant, allowing me to identify the effects of labor reallocation on manufacturing outcomes rather than any collective change in factors of production. In support of this premise, I find that increases in temperature have no effect on capital, management, or the entry of new plants.

These results suggest that the reallocation of labor across sectors could play an important role in attenuating the economic consequences of agricultural productivity shocks. Counterfactual estimates, examining the impact of temperature on total GDP, suggest that in the absence of labor reallocation total economic losses would be up to 40% larger. This highlights the role that liberalising goods and labor markets can play, as well as the importance of the local policy environment, in managing the economic consequences of weather-driven changes in agricultural productivity. In addition, I estimate that mitigating the adverse effects of temperature on manufacturing could offset the aggregate effects of temperature by up to 72% in the absence of any adaptation in the agricultural sector, suggesting that there

²Understanding whether the differences between casual and manufacturing workers in the manufacturing sector are driven by frictions or human capital differences is beyond the scope and capacity of the data and so remains an important question for future research.

could be considerable gains associated with managing the adverse effects of temperature in non-agricultural sectors. Collectively, these results provide insights into an important mechanism through which economic agents are able to manage climatic influence on economic outcomes, as well as highlighting the sensitivity of non-agricultural sectors to temperature increases.

The remainder of the paper is structured as follows: section 2 examines the relationship between weather and agricultural production; section 3 investigates the degree to which workers are able to move across sectors and space in response to weather-induced labor demand shocks; section 4 explores the impact of labor reallocation on manufacturing outcomes; section 5 discusses the implications of these results, considering the degree to which labor reallocation across sectors could offset losses to agriculture; section 6 concludes.

2 The Effects of Weather on Agricultural Markets

As in many developing countries, agriculture plays an important role in India's economy. During the time period of this study – the beginning of the 21st century –, agriculture accounted for roughly 15–20% of GDP, 60–70% of land use, and 40-50% of employment – many of whom are landless laborers employed on daily contracts.

A key feature of India's agricultural landscape is its dependence on the timing and intensity of the monsoon (Rosenzweig and Binswanger, 1993).³ Rainfall plays an important and salient role in the production of crops; however, the role of temperature, is a consideration often neglected in economic analysis. The monsoon's arrival in early summer is especially important for the kharif season, which corresponds with this period, but also for the rabi season, which begins at the end of the kharif season and continues through the cooler autumn and winter months before being harvested in the spring. Consequently, rabi yields are highly dependent on the degree to which rainfall can be stored in the soil. High temperatures prior to the monsoon affect the onset of the monsoon – a thermally driven phenomenon –, the degree to which rainfall drains from the soil, and soil temperature, which is important for seed germination and plant growth. High temperatures during the monsoon directly affect the kharif crop and increase the rate of evapotranspiration, which affects the availability of moisture in the soil, necessary for rabi crop production. Finally, high temperatures directly affect the rabi crop, even in the case in which irrigation is used.⁴

³During the period of study less than 30% of cultivated land is irrigated.

⁴While temperature is an important determinant of vapour pressure deficit, which irrigation can alleviate, around one third of the effects of temperature on yield losses arise due to an increase in the pace of crop development, which provides less time for the plant to develop and absorb nutrients and calories (Schlenker and Roberts, 2009). Fishman (2012) demonstrates these effects in the context of India by showing that

In this section I examine the effects of weather on two sets of agricultural outcomes. First, I examine the degree to which weather affects agricultural production in India, identifying the sign and magnitude of this relationship. Second, I examine the effects of weather on agricultural prices. This provides an insight into the expected response of labor following a change in agricultural productivity. A priori, it is ambiguous as to whether a reduction in agricultural productivity will result in an increase or decrease in the demand for labor. In a state of autarky, a reduction in agricultural production will result in an increase in prices as supply falls. Jayachandran (2006) shows that if workers have an inelastic labor supply and face a binding subsistence constraint for food (only relevant in the absence of trade), then a reduction in agricultural production will result in an increase in agricultural labor. Furthermore, an increase in prices could reduce the consumption base of the local economy, reducing demand for other commodities (Henderson et al., 2012, Soderbom and Rijkers, 2013, Santangelo, 2015, Emerick, 2016). By contrast, if the local economy is open to trade, then consumption and production are separable. In a state of autarky, agricultural surplus is necessary for the movement of workers into non-agricultural production, as the local economy is responsible for feeding itself. Only when enough food is produced can the economy focus on producing other products. However, when an economy is open to trade, the local economy does not need to produce food itself. Food can be imported and paid for by the export revenues of other commodities. Consequently, instead of rural prosperity fuelling the movement of workers out of agriculture – the historical norm for many developed countries –, rural deprivation pushes workers out of agriculture into other sectors of the economy. In the case of free trade, prices in the local economy are exogenous, set on the global market. Consequently, a local change in production will have a more muted affect on the price of tradable products if the locality is more open to trade, resulting in a change in local comparative advantage. Appendix A presents a simple model based on Matsuyama (1992) demonstrating how the comparative statics vary based on market integration. By understanding the responsiveness of prices to changes in the weather, we can gain an insight into the degree to which Indian districts are integrated into other markets, either national or international, allowing us to postulate the direction in which labor might move following a change in agricultural productivity. Section 3 will then test these insights directly using worker-level data on employment and wages.

higher temperatures still have a direct effect on rice yields – a crop known to be naturally resistant to higher temperatures – after controlling for irrigation.

2.1 Data – Yields and Prices

Data on crop yields and farm-gate prices come from the ICRISAT Village Dynamics in South Asia Macro-Meso Database (henceforth VDSA), which is a compiled from a number of official government data sources. The data analysed cover 12 major crops across 302 districts in 19 states between 1960 and 2009.⁵ For comparability with the other datasets I restrict my attention to the period 2001–2007. For each crop and district, the data provide the total area planted, total production in tonnes, and farm-gate prices. It is straightforward to calculate yields as total production divided by total area planted. I also calculate the value of production, defined as price multiplied by yield. Prices, by crop, are deflated to 2001 Rs. Panel A of Table 1 provides summary statistics for the VDSA data.

2.2 Data – Rainfall and Temperature

Rainfall and temperature data are collected from the ERA-Interim Reanalysis archive, which provides 6-hourly atmospheric variables for the period on a $0.25^{\circ} \times 0.25^{\circ}$ guadrilateral grid. Daily variables are calculated for each district centroid using inverse distance weighting from all grid points within 100km. The weight attributed to each grid point decreases guadratically with distance.⁶ Although India has a large system of weather stations that provide daily readings dating back to the 19th century, the spatial and temporal coverage of ground stations that report temperature and rainfall readings has sharply deteriorated over time. Furthermore, there are many missing values in the publicly available series. If we were to base the construction of this data on a selection rule that requires data for 365 days of the year, the database would have very few observations. Reanalysis data provides a solution to these issues and to endogeneity concerns related to the placement of weather stations, variation in the quality of data collection, and variation in the quantity of data collected. By combining observational data, from ground stations and remote-sensing products (satellites), with global climate models, reanalysis data provides a consistent best estimate of atmospheric parameters over time and space (Auffhammer et al., 2013). This results in an estimate of the climate system that is separated uniformly across a grid, that is more uniform in quality and realism that observations alone, and that is closer to the state of existence than any model could provide alone. This type of dataset is increasingly being used by economists, especially

⁵The 12 crops are Barley, Cotton, Finger Millet, Groundnut, Linseed, Maize, Pearl Millet, Rice, Rape and Mustard Seed, Sorghum, Sugarcane, and Wheat.

⁶The results are robust to alternative methods of construction, including: linear weights; cubic weights; the simple average of each point in the district; the average of each point in the district weighted by the area share of cultivated land; and the average of each point in the district weighted by population. Measures based on averages result in a smaller sample size, as some districts do not contain a data point and require the inverse distance weighting procedure.

in developing countries, where the quality and quantity of weather data is limited.⁷ Panel D of Table 1 provides summary statistics for the ERA-Interim Reanalysis Data.⁸

2.3 Empirical Specification – Yields and Prices

The unit of observation in this analysis is the crop \times district level.⁹ In 2001, the average district population was 1.75 million and the average area was 5,462 km² (Census of India, 2001).¹⁰ The main empirical specification for estimating the effect of weather on agricultural outcomes is based on the following model,

$$\log Y_{cdt} = f(w_{dt}) + \alpha_{cd} + \alpha_{ct} + \phi_s t + \varepsilon_{cdt}$$

where: Y_{cdt} represents the outcome of interest – yields, the value of production, or farmgate prices; α_{cd} is a vector of crop × district fixed effects; and α_{ct} is a vector of crop × year fixed effects, absorbing all unobserved time-varying differences in the dependent variable that are common across districts. However, the assumption that shocks or time-varying factors are common across districts is unlikely to be valid, so I also include a set of flexible, state-specific time trends, $\phi_s t$.

The last term is the stochastic error term, ε_{cdt} . I follow the approach of Hsiang (2010) by assuming that the error term ε_{dt} is heteroskedastic and serially correlated within a district over time (Newey and West, 1987) and spatially correlated across contemporaneous districts (Conley, 1999). For each result I loop over all possible distances up to 2000km, selecting the parameter value that maximises the standard errors. I then repeat this exercise for serial correlation, consistently resulting in a kernel of 1 year.¹¹

 $f(w_{dt})$ is a function of rainfall and temperature. In the most basic specification, $f(w_{dt})$ is modelled as a function of daily average temperature and total rainfall:

 $^{^{7}}$ All results are broadly robust to the use of alternative rainfall and temperature datasets from both satellite (TRMM) and ground station (UDEL) sources.

⁸Further details on all data sources are available in appendix B.

⁹Results are robust to aggregating across crops, using 2001 area weights, or to using the main crop in each district, defined using area planted in 2001.

¹⁰This is roughly twice the average area of a U.S. county (2,585 km²) and nearly 18 times greater than the average population of a U.S. county (100,000). When compared to commuting zones and labor market areas in the U.S. – developed because county boundaries are not considered adequate confines for an area's local economy and labor market –, Indian districts are approximately 4 times the population size (401,932) and around half the area (11,396 km²).

¹¹Results are also robust when standard errors are clustered at the state level. Fisher et al. (2012) report that clustering at the state level in the U.S. provides equivalent results to directly accounting for spatial correlation using the Conley (1999) standard error adjustment. The average state size in India, when compared to the United States, is roughly similar when compared to states east of the 100th meridian, the historic boundary between (primarily) irrigated and (primarily) rainfed agriculture in the United States.

$f(w_{dt}) = \beta_1(Temperature_{dt}) + \beta_2(Rainfall_{dt})$

As discussed, temperature is important for agricultural production both during and outside of the monsoon period. Consequently, I use crop calendars to define the relevant time period over which to construct the temperature variables. Alternative specifications, accounting for non-linearities in the temperature schedule are presented in Appendix E. Total rainfall is calculated for each state's monsoon period, beginning with the first month in which total monthly rainfall exceeds 100mm and ending with the first month that rainfall falls below 100mm.

2.4 Results – Yields and Prices

In Table 2 I estimate that a 1°C increase in temperature is associated with a 12.7% reduction in yield (column 1) and a 12.6% reduction in the value of production (column 2). In addition, a 100mm increase in rainfall is associated with a 1.15% increase in yield and a 1.07% increase in the value of production.

It is interesting to note that, in terms of its relative contribution, a one standard deviation change in temperature is shown to have a much larger effect on production (4.39%/SD) when compared to a one standard deviation change in rainfall (2.05%/SD), highlighting the important role that temperature plays in Indian agriculture.¹² This suggests that the importance attributed to rainfall for agricultural production in India may have been overestimated by the omission of temperature in previous work. Alternatively, it may have been the case that over time, farmers have become more effective in managing the effects of rainfall shocks, given the salient nature of the monsoon. This may also be due to the fact that rainfall is storable and can be substituted with ground water resources (manually, or through the use of irrigation systems), whereas the effects of temperature are more difficult to address, requiring heat-resistant crop varieties. The use of irrigation has been shown to offset the adverse effects of rainfall shortages; however, high temperatures still have a direct effect on yields even in the presence of irrigation (Fishman, 2012).

To consider the consequences of agricultural productivity shocks on labor demand, it is also of interest to understand the degree to which weather affects agricultural prices. In column 3 we observe that, on average, neither temperature or rainfall has a significant statistical or economic effect on agricultural prices.¹³ This suggests that Indian districts

 $^{^{12}{\}rm These}$ results are robust across weather data sets and over an extended period of analysis dating back to the 1960s.

 $^{^{13}}$ Allen and Atkin (2015) find a similar result looking at the effects of market access on agricultural prices in India between 1960 and 2010.

are reasonably well integrated with other markets, and can be considered as small, open economies. Consequently, a reduction in agricultural production should be associated with a reduction in the demand for agricultural labor, resulting in an outflow of workers into other tradable sectors of the economy due to a change in local comparative advantage. The next section formally tests this hypothesis.

3 The Effects of Weather on Employment, Wages and Migration

Given the significance of weather as a driver of short-run agricultural productivity, it is of interest to understand how these effects feed into labor market outcomes, providing insights into the consequences of weather shocks on the economic lives of agricultural workers. In this section I examine the effects of weather on wages, employment and unemployment within districts. In addition I explore the effects of weather on migration, examining the degree to which weather shocks in other districts affect employment outcomes in destination districts. This exercise provides insights into the relative importance of labor movements across vs. within districts in response to changes in agricultural productivity, as well as helping to bound local labor markets in India.

3.1 Data – Wages and Employment

Data on wages, employment and migration come from the National Sample Survey Organisation (hereafter, the NSS employment survey). The NSS employment survey is a nationally representative household survey which collects information on employment and wages in rural and urban areas. For the purpose of this analysis I make use of NSS survey rounds 60, 61, 62 and 64, covering 2003–04, 2004–05, 2005–06, and 2007–08. The level of analysis using the NSS data is at the district level. I restrict my attention to the sample of districts used in the analysis of agricultural yields, covering both rural and urban areas. I calculate the average day wage and the likelihood of being employed in each sector. The analysis focusses on four sectors, broadly defined as agriculture, manufacturing, services, and construction. The average daily wage is defined as the total wage received divided by the number of days worked over the previous seven days. The likelihood of being employed in each of the aggregated sector in a given district-year is calculated from individual responses to a survey question on their principal sector of engagement or whether they are unemployed, and provides district-level labor force employment share. Panel B of Table 1 provides summary statistics for wages and Panel C provides summary statistics for employment and unemployment shares. Agriculture accounts for an average of 44% of the labor force, with manufacturing employing 23%, services 18%, and construction 6%. Unemployment is is 8% of the labor force.¹⁴ Examining the differences in wages across sectors, we observe that the wage that agricultural laborers receive on average is significantly lower than the non-agricultural wage. Whether this unconditional wage gap is driven by adjustment costs, human capital differences, compensating differentials associated with sector-specific amenities, or bargaining power, is unclear; however, examining the degree to which workers are able to move across sectors in response to short-run productivity shocks provides some insight into the degree to which adjustment costs may be a first-order concern in this context.

3.2 Data – Migration

An important consideration is the degree to which workers may move across space, rather than sectors. Round 64 of the NSS Employment Survey contains a special schedule on seasonal migration. This provides data on the origin district of seasonal migrants; however, there is no detail on the destination of seasonal migrants. Instead, the NSS reports the destination of migrants in District ℓ_o in six relevant categories: rural or urban migration within the same District (m_{oo}) ; rural or urban migration between Districts in the same State $(\sum_{\ell_d \neq \ell_o \in S_o} m_{od})$; rural or urban migration between States $(\sum_{S_d \neq S_o} \sum_{\ell_d \neq \ell_o \in S_d} m_{od})$. Consequently, it is necessary to predict the district of destination for seasonal migrants who migrate to different districts. To do this, I draw inspiration from Imbert and Papp (2015) and use the 2001 Indian Population Census, extracting data on migrant workers by state of last residence. For each destination district, ℓ_d , I observe: the number of migrant workers from the same district (M_{dd}) ; the number of migrant workers from other districts in the same state $(\sum_{\ell_o \neq \ell_d \in S_d} M_{do})$; the number of migrant workers from districts in other states $(\sum_{S_o \neq S_d} \sum_{\ell_o \neq \ell_d \in S_o} M_{do})$. I combine these data to estimate seasonal migration flows \hat{m}_{od} , using the following algorithm:

$$\hat{m}_{od} = \begin{cases} m_{od} & \text{if } \ell_o = \ell_d \\ \frac{\sum_{\ell_o \neq \ell_d \in S_d} M_{do}}{\sum_{S_d} \sum_{\ell_o \neq \ell_d \in S_d} M_{do}} \sum_{\ell_d \neq \ell_o \in S_o} m_{od} & \text{if } \ell_o \neq \ell_d \text{ and } S_o = S_d \\ \frac{\sum_{S_o \neq S_d} \sum_{\ell_o \neq \ell_d \in S_o} M_{do}}{\sum_{S_d \geq S_o \neq S_d} \sum_{\ell_o \neq \ell_d \in S_o} M_{do}} \sum_{S_d \neq S_o} \sum_{\ell_d \neq \ell_o \in S_d} m_{od} & \text{if } \ell_o \neq \ell_d \text{ and } S_o \neq S_d \end{cases}$$

¹⁴Unfortunately, it is not possible to explore entry and exit from the labor force.

I deviate from Imbert and Papp (2015) in two respects. First, by using migrant workers rather than the total population of permanent migrants. Second, by broadening my attention beyond urban destinations. Non-agricultural production is not restricted to urban areas, and so rural-urban migration is not the appropriate characterisation of migration flows in the context of this paper. Indeed, a number of papers provide evidence to suggest that non-agricultural production in India is decentralising, from urban to peri-urban and even rural areas, taking advantage of cheaper labor and vastly cheaper land prices (Ghani et al., 2012, Desmet et al., 2015, Colmer, 2015). These adjustments provide stronger support for the identification assumption, on which this approach relies: that the proportion of NSS seasonal migrants who go from district ℓ_o to district ℓ_d , either in the same state or between states, is the same as the proportion of census migrant workers in district ℓ_d who come from another district ℓ_o , either in the same state or between states.

On average, rural-origin migrants comprise the bulk of migration flows, accounting for nearly 90% of all seasonal migration. 66.1% of migrants move within the same district, 2.6% of migrants move to another district within the same state (shared among an average of 15 districts per state, 0.17% per district), and 31.3% move to a different district in a different state (shared among an average of 577 districts, 0.05% per district).¹⁵ However, most strikingly, we observe that there is very little seasonal migration in absolute terms – only 1.1% of the population engage in seasonal migration. This is an observation that has been highlighted by a number of papers and contrasts starkly with migration patterns in other developing and developed countries (Foster and Rosenzweig, 2008, Munshi and Rosenzweig, 2015, Morten, 2013).

These insights have potential implications for the effects of localised shocks in India. First, if workers are limited in their ability to move across space, then the economic consequences of agricultural productivity shocks will be locally concentrated. Second, this implies that sectoral shocks are likely to have a bigger effect on other sectors in the local economy, as employment adjustments are less diversified across space. Finally, this implies that localised productivity shocks elsewhere are unlikely to have a large effect on economic outcomes across space; however, the validity of this argument is decreasing as the spatial correlation of localised productivity shocks increases, and as the importance of a specific location for the supply of workers increases. I test this prediction by examining the effects of localised temperature shocks in origin districts on employment and wages in destination districts to understand the degree to which localised productivity shocks propagate through labor markets across space.

 $^{^{15}593}$ districts - 1 state, i.e., an average of 16 districts = 577 out-of-state districts on average

3.3 Empirical Specification – Employment, Wages, and Migration

In analysing the effect of weather on employment, wages, and migration, the unit of analysis is the district level. The main empirical specification for estimating the effect of weather on local labor market outcomes is based on the following model,

$$Y_{dt} = f(w_{dt}) + \alpha_d + \alpha_t + \phi_s t + \varepsilon_{dt}$$

where: Y_{dt} represents the outcome of interest – sectoral labor force shares and the log of average wages; α_d is a vector of district fixed effects, absorbing all unobserved districtspecific time-invariant variation in the dependent variables; and α_t is a vector of year fixed effects, absorbing all unobserved time-varying differences in the dependent variable that are common across districts. I also include a set of flexible, state-specific time trends, $\phi_s t$.

As in the analysis on agricultural outcomes, $f(w_{dt})$ is a function of rainfall and temperature. In the most basic specification, $f(w_{dt})$ is modelled as a function of daily average temperature measured over the agricultural year, and total rainfall measured over the statespecific monsoon period. Alternative specifications, accounting for non-linearities in the temperature schedule are presented in Appendix E.

The last term is the stochastic error term, ε_{dt} . Standard errors are adjusted as in section 2.3.

The specification examining the degree to which weather-driven changes in agricultural productivity in "foreign" districts affect local labor market outcomes through migration differs slightly.

Using the bilateral migration flows discussed in section I construct a spatial weights matrix summarising the migratory relationship between each district. As mentioned, migration flows between ℓ_o and ℓ_d produce and $o \times d$ matrix $\mathbf{M}_{o \times d}$,

$$\mathbf{M}_{o\times d} = \begin{pmatrix} m_{11} & m_{12} & \cdots & m_{1D} \\ m_{21} & m_{22} & \cdots & m_{2D} \\ \vdots & \vdots & \ddots & \vdots \\ m_{D1} & m_{D2} & \cdots & m_{DD} \end{pmatrix}$$

Each weight m_{do} reflects the contribution of migration flows from district o to district d. In the case that all migration is spread equally between all districts, each entry in $M_{o\times d}$ will be equal to 1/d. At the other extreme, the case in which all migration occurs within districts provides an identity matrix. Based on the data, migration patterns in India tend towards the identity matrix extreme, far from an equal distribution of migrants.

To identify the degree to which local labor demand shocks affect economic outcomes

in destination sectors, I weight temperature and rainfall variation by the bilateral migration matrix, examining the migration-weighted effects of weather in district o on economic outcomes in district d through migration. The estimating equation is specified as follows,

$$Y_{dt} = \beta f(w_{dt}) + \gamma \left[\sum_{o} \frac{m_{od}}{M_d} \times f(w_{ot}) \right] + \alpha_d + \alpha_t + \phi_s t + \varepsilon_{dt}$$

where: Y_{dt} represents sectoral labor force shares in destination district d; α_d is a vector of district fixed effects; α_t is a vector of year fixed effects; $\phi_s t$ a set of state-specific time trends.

 $\sum_{o} \frac{m_{od}}{M_d} \times f(w_{ot})$ captures the migration-weighted effects of weather in other districts.

By directly controlling for local weather effects, $f(w_{dt})$, to account for the correlation of weather across space, γ identifies the effects of weather variation in foreign districts on local labor market outcomes through migration.

3.4 Results – Wages, Employment, and Migration

Table 3 presents the effects of temperature and rainfall on the average wage of workers in each sector within the local economy. A priori, the effect of weather on the average wage is ambiguous, as the overall effect depends on the change in composition of the workforce in each sector as well as the direct effects of temperature and rainfall changes. If, for a given level of demand, hot, dry weather reduces the supply of labor due to avoidance behavior then the average wage will rise. If, for a given level of supply, hot dry weather reduces the demand for labor there is less work available and so the average wage will fall. We observe that an increase in the daily average temperature is associated with a reduction in the average day wage for agricultural workers $(5.16\%/1^{\circ}C)$, consistent with a reduction in the demand for agricultural labor. As discussed, this could well be a function of both supply and demand forces, if workers are less willing to work in the heat, counteracting the reduction in the average wage. However, it is clear that the demand effect dominates. While this acts as an insurance mechanism for farm owners, a reduction in the average wage combined with a reduction in the availability of work – on the intensive or the extensive margin – could have significant welfare effects on agricultural workers if they are limited in their ability to find other work.

Interestingly, we see that rainfall limited effects on the average day wage. The sign on the estimated coefficient is negative and it is marginally significant. Furthermore, this effect is driven by an increase in the denominator (days worked in agriculture) rather than any change in the wage bill received. By contrast, temperature effects are driven by a reduction in the numerator, the wage bill. As discussed, rainfall is estimated to have less of an effect on agricultural production and so the impact on agriculture may not be significant enough to affect labor market outcomes. Associated with this consideration, it may simply be the case that, due to the relatively short time-series, there is not enough power to identify these effects.

In addition to the effects on agricultural wages, we observe that an increase in temperature is associated with a reduction in the average day wage in manufacturing $(3.4\%/1^{\circ}C)$; however this effect is not statistically significant. Again it is not clear as to how the average day wage in manufacturing should respond. In a simple model of labor reallocation, the movement of workers across sectors should reduce the average wage in the destination sectors. However, this depends on how the inflow of workers affects the wages of the incumbent workers. In the context of manufacturing where the tasks of workers is less uniform than in agriculture, there may be complementarities between worker types, which could increase the average wage in manufacturing. However, the limited movement in wages across other sectors, may simply indicate that workers are not moving across sectors. To understand the degree to which workers are moving, I estimate the effects of weather on employment and unemployment as shares of the labor force, identifying the degree to which workers are able to move across sectors, and find jobs, in response to reductions in the demand for agricultural labor.

Table 4 presents the results of this analysis. We observe that an increase in the daily average temperature is associated with a significant reduction in the district share of agricultural employment $(11\%/1^{\circ}C)$. Combined with the wage results, this indicates that temperature increases are associated with significant reductions in the demand for agricultural labor. The question remains as to whether these workers are able to find employment in other sectors of the economy, or whether they become unemployed. Consistent with the inferences drawn from the effects of weather on agricultural prices, we observe that the reduction in the share of agricultural employment is offset by an increase in the share of manufacturing employment $(7.61\%/1^{\circ}C)$ (and a smaller increase in the share of services employment $(3.86\%/1^{\circ}C)$). Of interest, we observe that there are no changes in unemployment, suggesting that workers in India are relatively unconstrained in their ability to move across sectors in response to transitory labor demand shocks.

Consistent with the premise that the effects of rainfall on agricultural production are not sufficient to drive labor market outcomes, rainfall is shown to have no significant effect on changes in the composition of employment in the local economy.¹⁶ This is consistent with the previous results, demonstrating that temperature has a relatively more important effect on agricultural production than rainfall – a premise that has found support in a number

¹⁶These results are robust across alternative weather data sets.

of other recent studies, emphasising the importance of temperature variation over rainfall as a driver of economic outcomes (Hsiang, 2010, Dell et al., 2012, Gray and Mueller, 2012, Burgess et al., 2014, Mueller et al., 2014, Burke et al., 2015).

In addition to looking at the effects of weather on local economic activity, I also examine the degree to which weather may affect labor market outcomes through migration. The purpose of this exercise is to examine whether short-run changes in the weather result in a reallocation of labor across space, distorting the definition of the local labor market and, consequently, the interpretation of the results, as well as being an outcome of interest in its own right. In particular, it is important to understand the degree to which changes in the population may distort changes in labor force shares. If increases in temperature results in out flows of workers then this would mechanically increase the employment share in the manufacturing and services sector. However, this would also mechanically increase the unemployment rate, even if there were no changes in unemployment.

Table 5 presents the results of this exercise. I find that the migration-weighted weather effects have no effect on employment shares in destination markets, indicating that there is little migration across districts in response to temperature increases. Consequently, local labor markets in India can be bounded at the district level. The reason behind the limited migration remains unclear: on the one hand workers may face significant adjustment costs across space; on the other hand, the ability of other sectors to absorb workers in response to sectoral productivity shocks somewhat mitigates the need to move across space.

Collectively, the results presented in this section suggest that workers in India are relatively able to move across sectors in response to transitory labor demand shocks, suggesting limited constraints on the supply-side. Furthermore, there appear to be limited demand-side constraints in response to these transitory labor demand shocks, with the manufacturing sector absorbing a significant share of these workers.

4 The Effects of Weather on Manufacturing

Having provided evidence to suggest that agricultural workers are relatively able to move across sectors in response to weather-induced changes in agricultural productivity and that the manufacturing sector is a major absorber of these workers, it is of interest to understand what these workers do and how they affect the productivity of firms and the labor market outcomes of incumbent workers. In turn these insights will help to shed light on the degree to which labor reallocation can attenuate the economic consequences of temperature increases.

4.1 Data – Manufacturing Plants

Data on manufacturing come from the Annual Survey of Industries (ASI) collected by the Ministry of Statistics and Program Implementation (MoSPI), Government of India. The ASI covers all registered industrial units that employ 10 or more workers and use electricity, or employ at least 20 workers and do not use electricity. The ASI frame is divided into two schedules: the census schedule, which is surveyed every year, and the sample schedule, which is randomly sampled every few years. The ASI has a much wider coverage than other datasets, such as the Census of Manufacturing Industries (CMI) and the Sample Survey of Manufacturing Industries (SSMI), and is comparable to manufacturing surveys in the United States and other industrialised countries. However, the ASI does not cover informal industry that falls outside the Factories Act of 1948. The formal sector accounts for approximately two-thirds of manufacturing output in India and is therefore not representative of all manufacturing activities. It is, however, representative of tradable manufacturing in India, since the informal sector trades very small volumes, if at all. Consistent with this premise Santangelo (2015) finds that there are no movements of workers into the informal manufacturing sector following rainfall-driven changes in agricultural productivity. Appendix B provides more details on the ASI data preparation. The sample used cover an average of 14,876 firms observed between 2001 and 2007, resulting in a total of 103,273 firm-year observations.

The outcomes of interest are the log of total output, employment, and the average day wage (defined as the total wage bill/the total number of man days worked during the year). Employment outcomes are examined for both permanent (non-managerial) workers and contract workers. The distinction between contract workers and permanent workers is important for this analysis, especially for regulated firms.¹⁷ Contract workers are on casual contracts and so a priori are the type of worker that one would expect to move between the agricultural and manufacturing sector.

Using worker-level data from the NSS (discussed in section 3), I estimate worker-level mincerian wage regressions to estimate the size of wage gaps after controlling for education, age, gender, district and year fixed effects. Table 6 shows that there is a significant wage gap between permanent manufacturing workers and agricultural workers, with permanent manufacturing workers earning 1.54 times more than agricultural workers, within local labor markets after controlling for individual characteristics.¹⁸ However, we observe that the average wage gap between casual manufacturing workers and agricultural workers

¹⁷The distinction between contract and permanent workers is less clear for unregulated firms. However, within unregulated firms we observe that contract workers earn less than permanent workers and so may be characterized as less skilled, or in more temporary/casual positions.

¹⁸This data does not make the distinction between the informal and formal sector.

almost disappears after controlling for individual characteristics, with casual manufacturing workers earning 9% more than agricultural workers. Consequently, there is greater common support between the wages of contract workers and agricultural workers, indicating that within low-skill groups workers are relatively substitutable across sectors. This suggests that labor markets in this context may not be dualistic across sectors per se (agriculture vs. non-agriculture), but rather can be characterised as dualistic across types of activities or skill. The fact that non-agricultural sectors contain workers that are more likely to engage in more skilled or productive activities conflates the interpretation of a dualistic labor market across sectors. However, a sectoral dimension may become more important as workers rise up the skill ladder and work in more specialized tasks, reducing the substitutability of workers across sectors.

In addition to the outcome variables described above, I construct two measures of productivity. The first is a simple measure: output per worker. While this is a crude measure of productivity, it provides a relatively useful measure of the average labor productivity of the firm. The second measure is an estimate of total factor productivity. Appendix B provides an explicit model of TFP, in the context of a profit-maximising firm, that I use to construct my empirical estimates.

4.2 The Labor Regulation Environment in India

The combination of manufacturing firm-level data with meteorological data provides the basis of this empirical analysis. However, this is not sufficient to identify how the movement of labor out of agriculture affects economic outcomes in the manufacturing sector. The key empirical challenge relates to the fact that, while exogenous changes in temperature are an important driver of short-run agricultural productivity, there are potentially many empirically relevant channels through which temperature could affect manufacturing outcomes. Consequently, any estimate of the reduced form estimate of temperature on the outcomes of interest will provide the net effect of all empirically relevant channels.

To try to address this challenge, I set out to identify the labor reallocation channel, net of the remaining empirically relevant channels, by exploiting variation in the propensity of firms to absorb workers in response to transitory weather shocks. To do this, I exploit a combination of spatial variation and firm-level exposure to India's labor regulation environment. In more flexible labor market environments, regulated firms should be relatively more able to absorb workers in response to transitory changes in labor demand, compared to firms in rigid labor market environments. Unregulated firms should not be directly affected by the regulation, and so there should be no differential effects of temperature on unregulated firms across labor regulation environments.¹⁹

Industrial regulation in India has largely been the result of central planning; however, the area of industrial relations is an exception to this, providing spatial variation in firms' incentives regarding the hiring and firing of workers following transitory changes in labor demand. The key piece of legislation used to measure state-level variation in sectoral mobility is the Industrial Disputes Act of 1947 (hereafter the IDA). The IDA regulates Indian Labor Law concerning trade unions, setting out conciliation, arbitration, and adjudication procedures to be followed in the case of an industrial dispute, and was designed to offer workers in the formal manufacturing sector protection against exploitation by employers. Up until the mid 1990s, the IDA was extensively amended at the state level, resulting in spatial variation in labor market rigidities. Besley and Burgess (2004) use these extensive state-level amendments (113 in total) to construct a measure of the labor regulation, environment studying its impact on manufacturing performance and urban poverty. By examining the amendments made in each state over time, states are coded as either neutral, pro-worker, or pro-employer. A pro-worker amendment is classified as one that decreases a firm's flexibility in the hiring and firing of workers; Pro-employer amendments are classified as increasing a firm's flexibility in hiring and firing. Importantly, one may be concerned that agricultural volatility or weather may have been correlated with the timing or the direction of amendments; however, fortunately I provide evidence to suggest that this isn't the case (see Appendix D).

The cumulation of these scores over time determines the state's labor regulation environment. Consequently, West Bengal, Maharashtra and Orissa are assigned as pro-worker states (rigid). Rajasthan, Tamil Nadu, Karnataka, Kerala and Andhra Pradesh are assigned as pro-employer states (flexible). The remaining states are assigned as neutral. This assignment captures spatial variation in the propensity of firms to take advantage of transitory labor supply changes arising from year-to-year changes in agricultural productivity.

However, state-level variation is not sufficient to identify the labor reallocation channel, as it may simply capture the heterogeneous effects of weather, general equilibrium effects, or other state-level variation, confounding the interpretation of the estimated coefficients.²⁰ I therefore combine this spatial variation with firm-level exposure to the regulation based on chapter 5b of the IDA, which specifies the size that firms can become before the IDA has a binding effect. The firm-size threshold is 50 in West Bengal, 300 in Uttar Pradesh, and 100 elsewhere.²¹ Consistent with these rule I demonstrate that that there is evidence

¹⁹In theory, unregulated firms may be affected indirectly through spillovers. For example, in rigid labor market environments there may be relatively increase in employment at unregulated firms, due to the limited labor market opportunities in the regulated sector.

²⁰Although, importantly there does not appear to be any common differential effect of temperature on non-manufacturing sectors across labor regulation environments (see Appendix D).

²¹Results are robust to applying a uniform threshold across all states away from the regulated threshold,

of bunching in the raw data just below the firm-size employment threshold for rigid labor market environments, but not for flexible labor market environments (see Appendix D).

A further consideration is whether the workers moving out of agriculture are likely to be affected by the IDA. A priori we would expect these workers to enter the regulated formal manufacturing sector as casual contract workers. This raises an important question about the degree to which the labor regulation environment impacts the employment of casual workers. Contract workers are not directly considered as workmen under the IDA and, consequently, are not de jure regulated within manufacturing firms. However, this does not mean that contract workers are not affected by the IDA (Bertrand et al., 2015, Chaurey, 2015). Contract workers are still de jure regulated by the IDA under the contractor that hires them. Consequently, the availability of these workers to firms in rigid labor market environments may be directly affected by the willingness of contractors to put these workers on the books in response to transitory changes in the weather. In addition, contract workers may be *de facto* affected by the IDA. On the one hand, the exemption of contract workers from the IDA may provide an added incentive to hire contract workers in rigid labor markets, allowing employers to bypass some of the regulations in the IDA. If so, this would imply that the labor reallocation channel would be relatively larger in rigid labor market environments. Looking at the data, one observes, consistent with this argument, that the share of firms using contract workers – an extensive margin measure – is higher in rigid markets than in flexible markets (see Figure 1). On the other hand, the use of contract workers has been vigorously, and in some cases violently, opposed by unions and permanent workers, suggesting that firms may face significant costs associated with hiring contract workers, especially in rigid labor market environments. Furthermore, the Contract Labor Regulation and Abolition Act of 1970 prohibits the use of contract labor if the work "is done ordinarily through regular workmen in that establishment." To the degree that this is enforced, this restricts the degree to which firms can bypass the IDA. Consequently, in rigid labor market environments, where it is expected that firms have to negotiate with unions over decisions that affect the labor force, the hiring of contract workers, in response to transitory changes in labor availability, may be restricted. In support of this premise we observe, on the intensive margin, that the share of workers employed as contract workers is higher in flexible labor market environments, suggesting that, conditional on hiring contract workers, firms in more flexible markets are able to hire more casual workers than firms in rigid labor markets (see Figure 1). Given that, on average, there is no difference in the total number of workers employed by firms across labor regulation environments, this implies that there is a higher proportion of contract workers in flexible than in rigid labor market environments (Table 7).

mitigating concerns that the results could be driven by the movement of firms around the size threshold.

In practice, whether there is a differential propensity to hire more casual workers in rigid or in flexible labor market environments is an empirical question. Most importantly, it does not affect the identification of the labor reallocation effect, which simply requires that there be a differential effect of temperature across labor regulation environments for regulated firms, but not unregulated firms.

Despite providing evidence of bunching in the firm-size distribution in rigid labor market environments, the fact that weather was not a driver of the amendments made to the IDA, and that the spatial variation in the IDA does not appear to have a differential effect on other unregulated sectors, threats to identification still remain if other factors that are correlated with temperature, the labor regulation environment, and outcome variables differ between regulated firms above the regulatory thresholds, but not unregulated firms below the regulatory threshold. This threat can be tested directly as below the regulatory threshold, there should be no direct differential impact of temperature on firms across labor regulation environments, and any differences that do arise help to sign the bias of the effect on regulated firms. The identification assumption is therefore that there are no other confounding factors across labor regulation environments that differentially affect firms above the regulatory firm-size threshold, but don't differentially affect firms below the regulatory firm-size threshold. An initial examination of this assumption through simple difference-in-means tests suggests that there do not appear to be any obvious differences in the characteristics of firms that would raise concerns (Table 7). The following section presents the empirical specification, allowing for a more formal test of this identification assumption.

4.3 Empirical Specification – Manufacturing Outcomes

To identify the sign and magnitude of the labor reallocation channel, I interact the net effects of weather with a measure of the labor regulation environment, splitting the sample at the regulatory firm-size threshold. The estimation equation for both samples is written as follows,

$$\log Y_{ijdt} = \beta f(w_{dt}) + \gamma f(w_{dt}) \times \text{FLEXIBILITY} + \alpha_{jd} + \alpha_{jt} + \phi_s t + \varepsilon_{ijdt}$$
(1)

The dependent variable, Y_{ijdt} , is the natural log of: total output (sales), employment (by worker type), the average day wage (by worker type), and various productivity measures. The unit of analysis is firm *i*, in sector *j*, in district *d*, at time *t*.

District × industry (α_{jd}) fixed effects absorb all unobserved time-invariant variation within these dimensions; industry × year (α_{jt}) fixed effects control for sector-specific timevarying differences in the dependent variable that are common across districts; and a set of flexible state-specific time trends ($\phi_s t$) relaxes the assumption that shocks or time-varying factors that affect the outcome variables are common across districts.

As in the previous sections, $f(w_{dt})$ is a function of rainfall and temperature,

$$f(w_{dt}) = \beta_1(Temperature_{dt}) + \beta_2(Rainfall_{dt})$$
(2)

where total rainfall is measured over the state-specific monsoon period and the daily average temperature is measured over the agricultural year. As for the previous sections, alternative specifications, accounting for non-linearities in the temperature schedule are presented in Appendix E.

The challenge associated with identifying the labor reallocation effect when estimating the simple linear regression model, absent the interaction term, is that β captures the sum of all empirically relevant channels through which temperature affects the manufacturing outcomes. This identifies the effects of temperature, but does not provide an economic interpretation.

The interaction term, $f(w_{dt}) \times$ FLEXIBILITY, captures the differential propensity of firms to absorb workers in response to increases in temperature. FLEXIBILITY is defined to allow for a continuous measure of the labor regulation environment, based on Besley and Burgess (2004), bounded between 0 and 1. West Bengal is the baseline state, coded 0 as it is the most rigid labor regulation environment. Andhra Pradesh and Tamil Nadu are coded as 1, as they are the most flexible labor regulation environments.

For regulated firms $\gamma_{regulated}$, identifies the labor reallocation effect, net of the remaining empirically relevant channels through which temperature effects manufacturing, β , if $\gamma_{unregulated} = 0$. Any deviations from this condition provide the sign and magnitude of the bias captured in $\gamma_{regulated}$ if $\gamma_{unregulated}$ is not zero. For example a positive differential effect on unregulated firms would suggest that the estimated differential effect on regulated firms would be upward biased. By contrast, a negative differential effect for unregulated firms would suggest that the estimated differential effect for unregulated firms would suggest that the estimated differential effect for unregulated firms would suggest that the estimated differential effect on regulated firms would suggest that the estimated differential effect on regulated firms would suggest that the estimated differential effect on regulated firms would suggest that the estimated differential effect on regulated firms would be downward biased.

The last term is the stochastic error term, ε_{dt} . Standard errors are adjusted as in section 2.3.

4.4 Results – Manufacturing

Unregulated Firms

I begin by examining the effects of temperature on unregulated firms. Below the regulatory threshold, there should be no direct differential impact of temperature on firms across labor regulation environments.²² As a result, these estimates do not disentangle the labor real-location effect, but rather test an important identification assumption: that any additional channels through which weather could affect manufacturing outcomes are constant across labor regulation environments. This also tests for the presence of any additional spatial differences such as general equilibrium effects or other policy differences that are correlated with the spatial dimension of the labor regulation environment.

Tables 8 and 9 present results that provide direct evidence in support of the identification assumption. Consistent with the results in section 3, I find evidence that an increase in temperature is associated with a net expansion of economic activity in unregulated firms (Panel A); however, there is limited evidence of a differential effect of temperature across labor regulation environments on unregulated firms.

The one exception to this is a differential effect in total employment. However, given the absence of other differential effects, it is plausible that this arises due to a spillover in the labor regulation environment itself, rather than differences in the other empirically relevant channels through which temperature affects manufacturing, or other policy or geographic differences. We observe that in more flexible markets there is relatively less hiring in response to temperature increases. This is consistent with the premise that the labor regulation environment may have indirect effects on unregulated firms. If workers are not able to find employment in the regulated sector of rigid labor market environments, we may expect a relative expansion (accounting for the size threshold) of the unregulated sector.

While it is not clear why firms require or distinguish between casual and permanent works in the unregulated market, it is interesting to note that the differential employment effects appear to be driven by contract workers, the type of workers that we might expect a priori to move in response to transitory changes in the demand for agricultural labor. For example, while regulatory constraints are not binding for unregulated firms it is plausible that firms may still have a preference to hire workers into temporary contract positions over permanent positions for other reasons, such as differential administrative costs.

These results suggest two things: 1) that our estimates of the differential effect of temperature are not likely to be biased due to confounding geographic or policy considerations that may be correlated with the labor regulation environment; 2) that any other empirically

 $^{^{22}}$ As discussed there may be indirect effects of the regulation on unregulated firms.

relevant channels through which temperature may affect manufacturing are constant across labor regulation environments. However, estimates of the differential effect of temperature on contract workers in regulated firms are likely to be a lower bound due to the potential spillover effects of the labor regulation environment onto unregulated firms.

Regulated Firms

In light of the evidence above, suggesting that the other empirically relevant effects of temperature are constant across labor regulation environments, and that other policy and geographic considerations do not appear to be confounding the interpretation of the labor regulation measure I proceed to explore the effects of temperature on regulated firms. Tables 10 and 11 present the results of this exercise.

First, I find evidence that there are limited net effects of temperature on manufacturing output and employment in regulated firms, as well as net reductions in productivity (Panel A). These results are consistent with, though not limited to, an expanding literature which suggests that high temperatures may have an adverse effect on labor productivity (Mackworth, 1946, 1947, Hsiang, 2010, Graff Zivin and Neidell, 2014, Adhvaryu et al., 2015, Graff Zivin et al., 2015, Somonathan et al., 2015). However, it is not clear whether the absence of employment and production effects are true zeros, or whether the labor reallocation effect observed on net in unregulated firms is confounded by other competing temperature effects.

In support of the latter interpretation I find differential effects of temperature across labor regulation environments. In rigid labor market environments I find that increases in temperature are associated with contractions in output, employment and productivity. By contrast, in flexible labor regulation environments we observe relative expansions in output and employment. This is consistent with the premise that firms in more flexible labor market environments have a greater capacity to absorb workers in response to weatherdriven changes in agricultural productivity. In addition, we observe that firms are hiring contract workers, with no net or differential change in the number of permanent workers, consistent with anecdotal evidence and a priori reasoning. Related to the discussion of the *de facto* impact of the labor regulation environment on contract workers, the relative increase in the employment of contract workers in more flexible labor markets, suggests that firms in more rigid labor markets may be incentivised against hiring contract workers, at least in response to short-run changes in the availability of workers.

From the workers' perspective, it is reasonable to suppose that agricultural workers on casual contracts would be more likely to find casual work in the manufacturing sector before moving into permanent work. In addition, the presence of centralized contractors that provide firms with casual labor significantly reduces search costs for these positions compared to permanent positions. This is consistent with the evidence provided by Bryan et al. (2014) in Bangladesh, Franklin (2015) in Ethiopia, and Hardy and McCasland (2015) in Ghana, who demonstrate that, in contexts without contractors, there are significant search costs associated with finding employment.

From the firms' perspective, the results are consistent with the premise that manufacturing firms hire workers on casual contracts as a screening process, rather than hiring movers into permanent contracts straight away. Employers face an adverse selection problem, as they can only discern a worker's true ability after a hiring decision has been made, especially in the absence of employment histories. By using contract workers, firms can learn more about a worker's productivity before deciding whether to hire them permanently. This is consistent with the evidence provided by Heath (2015) who finds that firms in Bangladeshi garment factories hire workers through referrals to mitigate adverse selection and moral hazard concerns. In doing so, firms can punish the referral provider if the new entrant is unproductive. Hardy and McCasland (2015) also highlight the importance of worker screening in the hiring decisions of firms in Ghana.

While there appears to be little impediment to moving across sectors within casual tasks, the absence of employment into permanent manufacturing positions suggests that casual and permanent labor markets are segmented, at least in the short run. Local labor markets in developing countries can therefore be characterised as dualistic, not in terms of sectors (agriculture vs. non-agriculture), but rather in terms of the type of employment in which workers engage (casual vs. permanent). This raises an interesting question about the degree to which casual workers face adjustment costs in the movement into permanent positions. As noted, there is a significant wage gap between casual manufacturing workers and permanent workers. However, while this gap exists, it is less clear how it should be interpreted. On the one hand, wage gaps may represent significant adjustment costs, implying that there are arbitrage opportunities to increase productivity if these costs could be reduced – a misallocation of talent (Banerjee and Duflo, 2007, Restuccia and Rogerson, 2008, Hsieh and Klenow, 2009, Moretti, 2011, Bryan et al., 2014, Gollin et al., 2014, Hsieh et al., 2014, Bandiera et al., 2015a, Bryan and Morten, 2015, Munshi and Rosenzweig, 2015). On the other hand, average wage gaps may simply represent differences in human capital between casual and permanent workers, with low-skilled workers selecting into casual tasks and highskilled workers selecting into permanent tasks (Roy, 1951, Heckman and Sedlacek, 1985, Heckman and Honore, 1990, Miguel and Hamory, 2009, Beegle et al., 2011, Lagakos and Waugh, 2013, Young, 2013, 2015). This interpretation would suggest that, while average wage gaps across sectors exist, marginal productivities may be equalised across activities – an efficient allocation of talent. Both of these channels may be further confounded by differences in the bargaining power or amenities across tasks. As discussed, the evidence presented so far, alongside evidence from worker-level mincerian wage regressions, suggests that adjustment costs, to the degree that they exist, are limited across sectors within casual activities.²³

In addition to the relative expansion of employment and production in flexible labor regulation environments, I also observe that temperature increases are associated with differential wage and productivity effects. Consistent with the relative inflow of contract workers in flexible labor regulation environments we observe a relative fall in the average wages of casual workers in response to temperature increases. Furthermore, we observe that temperature increases are associated with relative increases in the average wage of permanent workers. Given that contract and permanent labor markets are segmented, i.e., we observe no increase in the number of permanent workers, this suggests that the tasks that the casual entrants and permanent workers engage in are complementary in production. Consistent with this premise, we also observe relative increases in average labor productivity and measured TFPR, as well as the number of products produced.²⁴

A speculative interpretation of these findings is that the inflow of relatively low-skilled casual workers, freeing up permanent workers to engage in more productive tasks, moving firms down the average cost curve.

However, one concern may be that these effects are driven by accompanying changes in other factors of production, confounding this interpretation of the results. Yet, one of the attractive features of the empirical context and identification strategy is that the movement of workers across sectors is driven by short-run changes in the weather and so one may consider that other factors of production and the technology of the firm are held fixed. Table 12 directly tests this consideration. I begin by looking at the effects of temperature on capital and capital depreciation. If capital were to increase alongside labor, then it would be difficult to attribute increases in productivity and permanent worker wages to the reallocation of labor alone. Consistent with the premise that the other factors of production are held

²³Unfortunately, it is beyond the scope and capacity of the data to provide inferences about the relative contribution of these channels to the wage gap between casual and permanent manufacturing workers. However, in appendix F I provide an upper bound on the gains from reallocation, under the assumption that the total wage gap between casual and permanent workers is driven by adjustment costs. Understanding the relative importance of the role that adjustment costs play in impeding the movement of workers out of casual employment and into permanent positions remains an important area for future research.

²⁴TFPR (CES) allows for imperfect substitution between contract and permanent workers using exogenous estimates of the elasticity of substitution between contract and permanent workers combined with a nested CES production function (see Appendix C for details on estimation).

fixed, we observe that there is no change in capital or capital depreciation in response to temperature changes, and that this effect does not vary across labor regulation environments. Second, I consider the effects of temperature on the number of managers and the wages of managers. While a crude measure of the organisational structure of the firm, this provides some insights as to whether productivity increases could have been driven by organisational change or whether the increase in permanent worker wages could be driven by the extraction of rents from the firm. If this were the case then we may also expect managers to share in these rents. We observe neither an increase in the number of managers nor changes in the average wage of managers, suggesting that neither changes in management nor rent extraction appear to provide first-order explanations for the results. Finally, I explore whether the firm expands the number of plants -a proxy for entry and exit considerations that are not directly observable in the data. Again, we observe that the firm does not open or close plants in response to changes in temperature, suggesting that there are unlikely to be significant changes in the number of firms or in the market structure in response to changes in temperature. These findings suggest that the productivity and wage results can be interpreted as being driven by the increase in casual workers, rather than changes in other factors of production or changes in the technology or management structure of the firm.

In addition to the supporting evidence presented here, a number of additional robustness tests, including the removal of firms around the threshold, specification extensions that account for non-linearities in the temperature distribution, alternative definitions of the labor regulation environment and instrumental variable results, are presented in Appendices D and E.

The above results highlight the problems associated with the identification and interpretation of reduced-form weather results, but demonstrate the insights that can be gleaned from attempting to isolate specific channels and mechanisms through which weather can affect economic outcomes. I show that increases in temperature are associated with contractions of economic activity in rigid labor market environments, where firms are less able to absorb workers in response to weather-driven changes in agricultural productivity. However, in more flexible labor regulation environments I estimate that increase in temperature are associated with a relative expansion of manufacturing production and the employment of casual workers. However, unlike the effects of temperature on unregulated firms, this reallocation does not result in a net increase in output for regulated firms, as the adverse effects of temperature counteract the labor reallocation effect. While it is beyond the scope of this paper, and indeed the capacity of the data, to identify the precise mechanisms through which this residual net effect has an effect on manufacturing outcomes, one thing is clear: if these effects can be mitigated, the realised impact of temperature on manufacturing output through labor reallocation will be significantly larger, offsetting the economic losses associated with temperature increases in agriculture. The following section explores various counterfactuals, relating to the aggregate consequences of these mechansism.

5 Counterfactual Analysis

In this section I explore what my results imply for aggregate production in India. I consider two sets of counterfactual experiments. First, I consider the counterfactual impact associated with shutting down labor reallocation by increasing the rigidity of the labor market environment across India to the level of West Bengal (the most Pro-Worker State). Second, I consider the counterfactual gains associated with shutting down the adverse effects of temperature on manufacturing.

Baseline Effects

I begin by estimating the baseline effects of temperature on GDP, using data on sectoral GDP for each district, focusing on agriculture, manufacturing, construction and services. Table 13 presents the results of this exercise, showing that a 1°C increase in temperature is associated with a reduction in agricultural GDP (-11.6%/1°C), a reduction in total manufacturing GDP (-2.57%/1°C), and no change in services or construction GDP. Overall, a 1°C increase in temperature is associated with a 2.63% reduction in total GDP.²⁵

To explore various counterfactual environments I split total manufacturing GDP into three components: the informal manufacturing sector (34% of GDP), the regulated formal manufacturing sector (22% of GDP) and the unregulated formal manufacturing sector (44% of GDP). Taking as given the estimated effects of temperature on manufacturing output for the regulated and unregulated formal manufacturing sector (presented in Panel A of tables 8 and 10), the residual effect of temperature on the informal sector, necessary to induce a 2.57% reduction in total manufacturing GDP, is -18.3%. These baseline figures are reported in column 1 of Table 14.

Increasing the Rigidity of the Labor Market Environment

In columns 2, and 3 of Table 14 I consider the impact of increasing the rigidity of all labor market environments in India to the levels of West Bengal (the most Pro-Worker State). First, I consider the effects of increasing rigidity only in the regulated formal sector. I do this by inducing a 15.3% reduction in output for firms in the most flexible labor markets

²⁵The predicted effect on total GDP from aggregating the estimated effects of a 1 degree increase in temperature on other sectors, i.e., $\sum_{s} \beta_s \times GDP_s/GDP$ is -3.03%.

(Andhra Pradesh and Tamil Nadu), the effect decreasing in the rigidity of the labor market environment. As such, there is no change in output for West Bengal. In this counterfactual environment a 1°C increase in temperature would be associated with a 4.58% reduction in manufacturing GDP and a 3.31% reduction in total GDP, corresponding to a 9.24% increase in losses to total GDP. Secondly, I consider the effects of increasing the rigidity of the labor market environment in the unregulated formal sector, equivalent to increasing the scope of labor regulation to unregulated firms. In this counterfactual environment losses are even greater. A 1°C increase in temperature is associated with a 11.32% reduction in manufacturing GDP and a 4.25% reduction in total GDP, corresponding to a 40.26% increase in losses to total GDP. These simple counterfactuals highlight the importance of labor mobility, as well as the importance of the local policy environment, in managing the economic consequences of temperature increases.

Shutting Down the Adverse Effects of Temperature on Manufacturing Firms

Next, I consider how much of the losses to GDP could be offset by mitigating the adverse effects of temperature on manufacturing. For example, one could imagine that these effects could be attenuated through the use of cooling technologies.²⁶ Due to the competing effects of temperature on manufacturing, the net effect on regulated manufacturing firms is zero, and so labor reallocation only offsets the losses associated with the adverse effects of temperature for these firms. By setting the adverse effects of temperature to zero the labor reallocation effect will be positive, offsetting losses to agricultural GDP.²⁷

In column 4 I restrict my attention to regulated formal manufacturing firms. In this case, a 1°C increase in temperature is associated with a 1.01% increase in manufacturing GDP and a 2.53% reduction in total GDP. This corresponds to a 16.5% reduction in losses to total GDP.

In column 5, I allow the estimated effects to be extrapolated to the rest of the formal manufacturing sector, expanding "cooling technologies" to unregulated formal manufacturing firms. In this counterfactual a 1°C increase in temperature is associated with a 7.74% increase in manufacturing GDP, and a 1.59% reduction in total GDP. This corresponds to a 47.52% reduction in losses to total GDP.

Finally, in column 6, I allow the estimated effects to be further extrapolated to the

 $^{^{26}{\}rm These}$ counterfactual exercises do not address the costs associated with the implementation of such technologies.

²⁷One concern relating to the validity of this exercise is that firms may only be hiring workers in response to the reductions in productivity associated with the adverse effects of temperature. Consequently, in the absence of adverse temperature effects, firms may have limited capacity to expand production in the short run. However, the net increase in output and employment for unregulated firms suggest that expansions in response to year-to-year variations in temperature are possible.

informal sector. I estimate that a 1°C increase in temperature is associated with a 12.95% increase in manufacturing GDP and a 0.86% reduction in total GDP. This corresponds to a 71.67% offset in losses to total GDP.

These results suggest that there could be significant gains from mitigating the adverse effects of temperature on manufacturing, and that the movement of workers across sectors could significantly offset the aggregate effects of temperature on local economic activity.

6 Conclusion

One of the salient features of economic life in developing countries is the centrality of agriculture to employment. Consequently, given the inextricable link between agricultural production and the environment, understanding the relationship between economic and natural systems can provide important insights into the economics lives of the poor. While we have a reasonably good understanding about how weather affects agricultural markets, understanding how workers respond to changes in labor demand helps to provide insights into the mechanisms through which climatic influence affects economic outcomes, as well as the functioning of labor markets in developing countries.

Consistent with a large literature examining the effects of weather on agricultural production, I estimate that temperature is a strong driver of short-run agricultural productivity. However, I also estimate that there are no effects on agricultural prices, consistent with a "law of one price", indicating that Indian districts are reasonably well integrated with other markets. A priori, this suggests that reductions in agricultural productivity should result in an outflow of workers into other sectors due to a local change in comparative advantage.

Consistent with this premise, I present evidence to suggest that agricultural workers in India are relatively able to move across sectors within local labor markets when temperature increases, moving chiefly into the manufacturing sector. This movement completely offsets the reduction in agricultural employment, with no increases in unemployment, or population through migration, indicating that the ability of other sectors to absorb workers is a key channel through which workers can manage reductions in the demand for their labor in agriculture. These results highlight the role that market integration and diversification can play in attenuating the aggregate consequences of sectoral productivity shocks.

In light of these results, I explore how this reallocation of labor across sectors affects economic outcomes in the formal manufacturing sector, which is representative of tradable industry in India. The principal challenge associated with identifying the effects of labor reallocation is that there are many channels through which temperature could affect manufacturing outcomes. Consequently, the estimated effect of temperature on manufacturing outcomes provides a net effect of all the empirically relevant channels, without a clear economic interpretation.

To discern the impact of labor reallocation on these firms, I interact the net effects of temperature with a combination of spatial and firm-level exposure to India's labor regulation environment, providing variation in the propensity of firms to absorb labor in response to short-run changes in labor availability.

For unregulated firms I estimate that the net effect of temperature on production and employment is positive and significant consistent with the premise that workers are able to move across sectors in response to weather-driven changes in agricultural productivity. However, importantly for identification, I estimate that there are no differences in the effects of temperature across labor regulation environments, indicating that any additional channels, through which temperature affects manufacturing, are constant across labor regulation environments. By contrast, for regulated firms I estimate a differential effect across labor regulation markets. In rigid labor market environments, an increase in temperature is associated with a contraction in economic activity, consistent with a literature that suggests that temperature is an important determinant of labor productivity. However, I demonstrate that there is a relative expansion of output and the employment of low-skilled casual workers in flexible labor regulation environments, consistent with the premise that regulated firms in these environments have a greater propensity to absorb workers in response to weather-driven changes in agricultural productivity. These results support the premise that the ability of firms to absorb workers is a key channel through which workers are able to manage agricultural productivity shocks, indicating that the local policy environment can play an important role by affecting the ability of firms to absorb labor.

Furthermore, I estimate that this inflow of casual workers is associated with an increase in the wages of permanent workers as well as increases in productivity, suggesting that there may be complementarities between the tasks that casual and permanent workers engage in. In support of this interpretation, I find that there are no changes in capital, management, or the number of plants, suggesting that the effects are driven by labor reallocation, rather than changes in other factors of production, or the technology or organization of the firm.

In considering the aggregate consequences of these effects, I explore two sets of counterfactual exercises using data on district level, sectoral GDP. First, I estimate that in the absence of labor reallocation total economic losses would be up to 40% larger, highlighting the importance of labor mobility in attenuating the economic consequences of sectoral productivity shocks. Second, I estimate that attenuating the adverse effects of temperature on manufacturing could offset the aggregate effects of temperature by up to 72%, despite the adverse effects that temperature increases have on the agricultural sector. This suggests that there could be considerable gains to managing the economic consequences of temperature in non-agricultural sectors.

The findings of this paper have three main implications. First, regarding the labor market decisions of the poor, my results suggest that workers in agriculture are highly responsive to changes in the agricultural wage and employment opportunities, resulting in movements across sectors within casual employment activities. These findings suggest that low-skilled workers are relatively substitutable across sectors. However the absence of movement into permanent manufacturing positions suggests that labor markets can still be characterised as dualistic (Lewis, 1954). However, labor markets do not appear to be dualistic across sectors (agriculture vs. non-agriculture) but rather in terms of the type of employment activities in which workers engage (casual vs. permanent). Consequently, when engaged in casual employment, the delineation of activities by sector may have little relevance, with workers engaging in activities across sectors in rural or urban areas of the local labor market. However, as workers move up the skill ladder into permanent jobs, the delineation of employment by sector may start to become more important. Many important research opportunities remain to help improve our understanding of whether workers face constraints that impede their movement out of casual and into permanent employment, and whether these constraints are amenable to policy.

Second, regarding the behaviour of firms, my results suggest that firms in India have the potential to act as a major absorbers of labor, even in the short-run, highlighting the importance of diversification in the management of idiosyncratic productivity shocks. In addition, I demonstrate that even sectors considered to be considerably less climate-sensitive than agriculture can be significantly affected by temperature increases, suggesting that we may significantly underestimate the damages associated with future climate change if we fail to account for non-agricultural impacts. Furthermore, in the face of competing mechanisms we will underestimate the economic importance of these damages. Understanding the relationship between environmental conditions and firm behaviour remains a fruitful area of research; especially questions relating to how the management and innovation of firms may help to manage short- and long-run environmental change.

Finally, regarding climatic influence on economic outcomes, my results show that workers are relatively able to adapt to temperature increases by moving across sectors, and that the ability of firms to absorb these movements is a key channel through which workers are able to manage the effects of weather-driven changes in agricultural productivity. Consequently, we may overestimate the damages associated with future climate change if we do not take into account the adaptation responses of economic agents. Much more work is required to understand how different institutions, systems, technologies, and policies may moderate the short-run and long-run environmental change so that we are better able to understand the constraints that economic agents face in managing such change. In turn we can better design and implement policy, where necessary, to mitigate the economic consequences of environmental change, today and in the future.

References

- ACEMOGLU, A., V. CARVALHO, A. OZDALGAR, AND A. TAHBAZ-SALEHI (2012): "The Network Origins of Aggregate Fluctuations," *Econometrica*, 80, 1977–2016.
- ADHVARYU, A., A. CHARI, AND S. SHARMA (2013): "Firing Costs and Flexibility: Evidence from Firms' Labor Adjustments to Shocks in India," *Review of Economics and Statistics*, 95, 725–740.
- ADHVARYU, A., N. KALA, AND A. NYSHADHAM (2015): "The Light and the Heat: Productivity Co-benefits of Energy-saving Technology," *Mimeo*.
- AHSAN, A. AND C. PAGÉS (2009): "Are All Labor Regulations Equal? Evidence from Indian Manufacturing," *Journal of Comparative Economics*, 37, 62–75.
- ALCOTT, H., A. COLLARD WEXLER, AND S. O'CONNEL (2015): "How Do Electricity Shortages Affect Industry? Evidence from India," *American Economic Review*.
- ALLEN, D. AND D. ATKIN (2015): "Volatility, Insurance, and the Gains from Trade," *Mimeo*.
- AMIRAPU, A. AND M. GECHTER (2014): "Indian Labour Regulations and the Cost of Corruption: Evidence from the Firm Size Distribution," *Mimeo*.
- ASHER, S. AND P. NOVOSAD (2014): "The Impact of Agricultural Output on Local Economic Activity in India," *Mimeo*.
- AUFFHAMMER, M., S. HSIANG, W. SCHLENKER, AND A. SOBEL (2013): "Using Weather Data and Climate Model Output in Economic Analyses of Climate Change," *Review of Environmental Economics and Policy*.
- AUTOR, D., D. DORN, AND G. HANSON (2013): "The China Syndrome: Local Labor Market Effects of Import Competition in the United States," *American Economic Review*.
- BANDIERA, O., R. BURGESS, N. DAS, S. GULESCI, I. RASUL, AND M. SULAIMAN (2015a): "The Misallocation of Labor in Village Economies," *Mimeo*.

- BANDIERA, O., A. PRAT, AND R. SADUN (2015b): "Managing the Family Firm: Evidence from CEOs at Work," *Mimeo*.
- BANERJEE, A. AND E. DUFLO (2007): "The Economic Lives of the Poor," Journal of Economic Perspectives, 21, 141–168.
- BEEGLE, K., J. DE WEERDT, AND S. DERCON (2011): "Migration and Economic Mobility in Tanzania: Evidence from a Tracking Survey," *Review of Economics and Statistics*, 93, 1010–1033.
- BENTOLILA, S. AND G. BERTOLA (1990): "Firing Costs and Labour Demand: How Bad is Eurosclerosis," *The Review of Economic Studies*, 57, 381–402.
- BERTRAND, M., C. HSIEH, AND N. TSIVANIDIS (2015): "Contract Labor and Firm Growth in India," *Mimeo*.
- BESLEY, T. AND R. BURGESS (2004): "Can Labor Regulation Hinder Economic Performance? Evidence from India," *Quarterly Journal of Economics*, 19, 91–134.
- BRYAN, G., S. CHOWDHURY, AND A. MOBARAK (2014): "Underinvestment in a Profitable Technology: The Case of Seasonal Migration in Bangladesh," *Econometrica*, 82, 1671– 1758.
- BRYAN, G. AND M. MORTEN (2015): "Economic Development and the Spatial Allocation of Labor: Evidence from Indonesia," *Mimeo*.
- BURGESS, R., O. DESCHENES, D. DONALDSON, AND M. GREENSTONE (2014): "The Unequal Effects of Weather and Climate Change: Evidence from Mortality in India," *Mimeo*.
- BURGESS, R. AND D. DONALDSON (2010): "Can Openness Mitigate the Effects of Weather Shocks? Evidence from India's Famine Era," *American Economic Review: Paper and Proceedings*, 100, 449–453.
- (2012): "Railroads and the Demise of Famine in Colonial India," *Mimeo*.
- BURKE, M., S. HSIANG, AND E. MIGUEL (2015): "Global Non-linear Effect of Temperature on Economic Production," *Nature*.
- BUSTOS, P., B. CAPRETTINI, AND J. PONTICELLI (2015): "Agricultural Productivity and Structural Transformation: Evidence from Brazil," *Mimeo*.
- CACHON, G., S. GALLINO, AND M. OLIVARES (2012): "Severe Weather and Automobile Assembly Productivity," *Mimeo*.
- CHAUREY, R. (2015): "Labor Regulations and Contract Labor Use: Evidence from Indian Firms," *Journal of Development Economics*, 114, 224–232.
- COLMER, J. (2015): "Urbanization, Growth, and Development: Evidence from India," *Mimeo*.
- CONLEY, T. (1999): "GMM Estimation with Cross Sectional Dependence," Journal of Econometrics, 92, 1–45.
- COSTINOT, A., D. DONALDSON, AND C. SMITH (2015): "Evolving Comparative Advantage and the Impact of Climate Change in Agricultural Markets: Evidence from a 1.7 Million Fields around the World." *Journal of Political Economy*, forthcoming.
- DEATON, A. (1992): "Saving and Income Smoothing in Côte D'Ivoire," Journal of African Economies, 1, 1–24.
- DELL, M., B. JONES, AND B. OLKEN (2012): "Temperature Shocks and Economic Growth: Evidence from the Last Half Century," *American Economic Journal: Macroeconomics*, 4, 66–95.
- DESMET, K., E. GHANI, S. O'CONNEL, AND E. ROSSI-HANSBERG (2015): "The Spatial Development of India," *Journal of Regional Science*.
- DONALDSON, D. (2015a): "The Gains from Market Integration," Annual Review of Economics, forthcoming.
- (2015b): "Railroads of the Raj: Estimating the Impact of Transportation Infrastructure," *American Economic Review*, forthcoming.
- EMERICK, K. (2016): "Agricultural productivity and the sectoral reallocation of labor in rural India Agricultural productivity and the sectoral reallocation of labor in rural India Agricultural Productivity and the Sectoral Reallocation of Labor in Rural India," *Mimeo.*
- FISHER, A., M. HANEMANN, M. ROBERTS, AND W. SCHLENKER (2012): "The Economic Impacts of Climate Change: Evidence from Agricultural Output and Random Fluctuations in Weather: Comment," *American Economic Review*, 102, 3749–60.
- FISHMAN, R. (2012): "Climate Change, Rainfall Variability and Adaptation through Irrigation: Evidence from Indian Agriculture," *Mimeo*.

- FOSTER, A. AND M. ROSENZWEIG (2004): "Agricultural Productivity Growth, Rural Economic Diversity, and Economic Reforms: India 1970 - 2000," *Economic Development and Cultural Change*, 52, 509–542.
- (2008): "Economic Development and the Decline of Agricultural Economic Development and the Decline of Agricultural Employment," *Handbook of Development Economics*, 4.
- FRANKLIN, S. (2015): "Location, Search Costs, and Youth Unemployment: A Randomized Trial of Transport Subsidies in Ethiopia," *Mimeo*.
- GHANI, E., A. GOSWAMI, AND W. KERR (2012): "Is India's Manufacturing Sector Moving Away from Cities?" *NBER Working paper 17992*.
- GOLLIN, D., D. LAGAKOS, AND M. WAUGH (2014): "The Agricultural Productivity Gap," Quarterly Journal of Economics, 129, 939–993.
- GRAFF ZIVIN, J., S. HSIANG, AND M. NEIDELL (2015): "Temperature and Human Capital in the Short- and Long-Run," *NBER Working Paper 21157*.
- GRAFF ZIVIN, J. AND M. NEIDELL (2014): "Temperature and the allocation of time: Implications for climate change," *Journal of Labour Economics*, 32, 1–26.
- GRAY, C. AND V. MUELLER (2012): "Natural disasters and population mobility in Bangladesh," *Proceedings of the National Academy of Sciences.*
- GUITERAS, R. (2009): "The Impact of Climate Change on Indian Agriculture," Mimeo.
- HALTIWANGER, J., S. SCARPETTA, AND H. SCHWEIGER (2008): "Assessing Job Flows Across Countries: The Role of Industry, Firm Size, and Regulations," *NBER Working Paper No. 13920.*
- HAMERMESH, D. (1993): Labour Demand, Princeton University Press.
- HARDY, M. AND J. MCCASLAND (2015): "Are Small Firms Labor Constrained? Experimental Evidence from Ghana," *Mimeo*.
- HEAL, G. AND J. PARK (2014): "Feeling the Heat: Temperature, Physiology, and the Wealth of Nations," *Mimeo*.
- HEATH, R. (2015): "Why Do Firms Hire Using Referrals? Evidence from Bangladeshi Garment Factories," *Mimeo*.

- HECKMAN, J. AND B. HONORE (1990): "The Empirical Content of the Roy Model," *Econo*metrica, 58, 1121–1149.
- HECKMAN, J. AND G. SEDLACEK (1985): "Heterogeneity, Aggregation, and Market Wage Functions: An Empirical Model of Self-Selection in the Labor Market," *Journal of Political Economy*, 93, 1077–1125.
- HECKMAN, J. A. (2003): "Law and Employment: Lessons from Latin American and the Caribbean," *NBER Working Paper No. 10129*.
- HENDERSON, J., A. STOREYGARD, AND U. DEICHMAN (2015): "Has Climate Change Driven Urbanization in Africa?" *Mimeo*.
- HENDERSON, J., A. STOREYGARD, AND D. WEIL (2012): "Measuring Economic Growth from Outer Space," *American Economic Review*, 102, 994–1028.
- HORNBECK, R. (2012): "The Enduring Impact of the American Dust Bowl: Short- and Long-run Adjustments to Environmental Catastrophe," *American Economic Review*.
- HORNBECK, R. AND P. KESKIN (2015): "Does Agriculture Generate Local Economic Spillovers? Short-Run and Long-Run Evidence from the Ogallala Aquifer," *American Economic Journal: Economic Policy*, 7, 192–213.
- HORNBECK, R. AND E. MORETTI (2015): "Who Benefits from Productivity Growth? The Local and Aggregate Impacts of Local TFP Shocks on Wages, Rents, and Inequality," *Mimeo*.
- HORNBECK, R. AND S. NAIDU (2014): "When the Levee Breaks: Black Migration and Economic Development in the American South," *American Economic Review*, 104, 963–990.
- HSIANG, S. (2010): "Temperatures and cyclones strongly associated with economic production in the Caribbean and Central America." Proceedings of the National Academy of Sciences, 107, 15367–15372.
- HSIEH, C., E. HURST, C. JONES, AND P. KLENOW (2014): "The Allocation of Talent and US Economic Growth," *Mimeo*.
- HSIEH, C. AND P. KLENOW (2009): "Misallocation and Manufacturing TFP in China and India," *Quarterly Journal of Economics*, 124, 1403–1448.

- IMBERT, C. AND J. PAPP (2015): "Short-term Migration, Rural Workfare Programs and Urban Labor Markets: Evidence from India," *Mimeo*.
- JAYACHANDRAN, S. (2006): "Selling Labor Low: Wage Responses to Productivity Shocks in Developing Countries," *Journal of Political Economy*, 114, 538–575.
- JINA, A. AND S. HSIANG (2015): "Geography, Depreciation, and Growth," American Economic Review: Paper and Proceedings.
- KALA, N. (2015): "Ambiguity Aversion and Learning in a Changing World: The Potential Effects of Climate Change From Indian Agriculture," *Mimeo.*
- KAUR, S. (2014): "Nominal Wage Rigidity in Village Labor Markets," Mimeo.
- KENRICK, D. AND S. MCFARLANE (1986): "Ambient Temperature and Horn Honking: A Field Study of the Heat/Aggression Relationship," *Environment and Behaviour*, 18, 179–191.
- KESKIN, P. (2010): "Thirsty Factories, Hungry Farmers: Intersectoral Impacts of Industrial Water Demand," *Mimeo*.
- LAGAKOS, D. AND M. WAUGH (2013): "Selection, Agriculture, and Cross-Country Productivity Differences," *American Economic Review*, 103, 948–980.
- LEWIS, W. (1954): "Economic Development with Unlimited Supplies of Labor," Manchester School of Economic and Social Studies, 22, 139–191.
- MACKWORTH, N. (1946): "Effects of Heat on Wireless Telegraphy Operators Hearing and Recording Morse Messages," *British Journal of Industrial Medicine*, 3, 143–158.
- (1947): "High Incentives versus Hot and Humid Atmospheres in a Physical Effort Task," *British Journal of Psychology: General Section*, 38, 90–102.
- MARDEN, S. (2015): "The Agricultural Roots of Industrial Development: Forward Linkages in Reform Era China," *Mimeo*.
- MATSUYAMA, K. (1992): "Agricultural Productivity, Comparative Advantage, and Economic Growth," *Journal of Economic Theory*, 58, 317–334.
- MIAN, A. AND A. SUFI (2015): "What Explains the 2007-2009 Drop in Employment?" *Econometrica*, forthcoming.

- MIGUEL, E. AND J. HAMORY (2009): "Individual Ability and Selection into Migration in Kenya," UNDP, Human Development Research Paper 2009/45.
- MOBARAK, A. AND M. ROSENZWEIG (2014): "Risk, Insurance and Wages in General Equilibrium," *Mimeo*.
- MORETTI, E. (2011): "Local Labor Markets," in *Handbook of Labor Economics*, ed. by D. Card and O. Ashenfelter, Elsevier.
- MORTEN, M. (2013): "Temporary Migration and Endogenous Risk Sharing in Village India," *Mimeo*.
- MUELLER, V., C. GRAY, AND K. KOSEC (2014): "Heat stress increases long-term human migration in rural Pakistan," *Nature Climate Change*.
- MUNSHI, K. AND M. ROSENZWEIG (2015): "Networks and Misallocation: Insurance, Migration, and the Rural-Urban Wage Gap," *Mimeo*.
- NEWEY, W. AND K. WEST (1987): "A Simple, Positive Semi-Definite, Heteroskedasticity and Autocorrelation Consistent Covariance Matrix," *Econometrica*, 55, 703–708.
- NICKELL, S. (1978): "Fixed Costs, Employment, and Labour Demand over the Cycle," *Economica*, 45, 329–345.
- OI, W. (1962): "Labour as a Quasi-Fixed Factor," The Journal of Political Economy.
- PAXSON, C. (1992): "Using Weather Variability to Estimate the Response of Savings to Transitory Income in Thailand," *American Economic Review*, 82, 15–33.
- RESTUCCIA, D. AND R. ROGERSON (2008): "Policy Distortions and Aggregate Productivity with Heterogenous Plants," *Review of Economy Dynamics*, 11, 707–720.
- ROSENZWEIG, M. AND H. BINSWANGER (1993): "Wealth, Weather Risk, and the Profitability of Agricultural Investment," *Economic Journal*, 103, 56–78.
- ROY, A. (1951): "Some Thoughts on the Distribution of Earnings," Oxford Economic Papers, 3, 135–146.
- RYAN, N. (2014): "The Competitive Effects of Transmission Infrastructure in the Indian Electricity Market," *Mimeo*.
- SANTANGELO, G. (2015): "Firms and Farms: The Impact of Agricultural Productivity on the Local Indian Economy," *Mimeo*.

- SCHLENKER, W. AND M. ROBERTS (2009): "Nonlinear temperature effects indicate severe damages to U.S. crop yields under climate change," *Proceedings of the National Academy* of Sciences, 106, 15594–15598.
- SODERBOM, M. AND B. RIJKERS (2013): "The Effects of Risk and Shocks on Non-Farm Enterprise Development in Rural Ethiopia," *World Development*, 45, 119–136.
- SOMONATHAN, E., R. SOMONATHAN, A. SUDARSHAN, AND M. TEWARI (2015): "The Impact of Temperature on Productivity and Labor Supply: Evidence from Indian Manufacturing," *Mimeo*.
- TARAZ, V. (2012): "Adaptation to Climate Change: Historical Evidence from the Indian Monsoon," Mimeo.
- TOWNSEND, R. (1994): "Risk and Insurance in Village India," Econometrica, 62, 539–592.
- VOLLRATH, D. (2009): "How Important are Dual Economy Effects for Aggregate Productivity," Journal of Development Economics, 88, 325–334.
- (2014): "The Efficiency of Human Capital Allocations in Developing Countries," Journal of Development Economics, 108, 106–118.
- YOUNG, A. (2013): "Inequality, the Urban-Rural Gap and Migration," *Quarterly Journal* of *Economics*, 128, 1727–1785.
- (2015): "Structural Transformation, the Mismeasurement of Productivity Growth, and the Cost Disease of Services," *American Economic Review*, forthcoming.

Figures and Tables

	Mean	STD. DEV. (within)	STD. DEV. (between)
Panel A: Agricultural Data			
YIELD	1.789	0.472	1.683
VALUE (Rs.)	$19,\!300.67$	10,392.71	$22,\!245.56$
Production ('000 Tonnes)	113.042	49.626	252.894
Area ('000 Hectares)	59.261	15.051	100.687
PRICE (Rs./Tonne)	$11,\!995.1$	4,012.233	7,431.825
Number of Crops	7.047	0	3,532
Average Crop Share	0.154	0.0276	0.216
Average Share of Main Crop	0.559	0.042	0.176
Panel B: Wage Data			
Average Day Wage: Agriculture	56.482	15.822	21.988
AVERAGE DAY WAGE: MANUFACTURING	96.720	42.627	37.397
Average Day Wage: Services	186.403	45.826	37.786
AVERAGE DAY WAGE: CONSTRUCTION	81.458	33.010	29.123
Panel C: Employment Data			
DISTRICT EMPLOYMENT SHARE: AGRICULTURE	0.445	0.095	0.147
District Employment Share: Manufacturing	0.225	0.062	0.086
DISTRICT EMPLOYMENT SHARE: SERVICES	0.183	0.046	0.057
DISTRICT EMPLOYMENT SHARE: CONSTRUCTION	0.066	0.031	0.032
UNEMPLOYMENT SHARE OF LABOR FORCE	0.078	0.032	0.043
Panel D: Meteorological Data			
Daily Average Temperature (°C)	24.847	0.271	4.185
Degree Days $(t_L = 17, t_H = \infty)$	3,103.204	90.757	809.660
DEGREE DAYS $(t_L = 0, t_H = 17)$	5,995.568	22.590	704.792
Monsoon Rainfall (mm)	927.297	206.509	482.657

Table 1: Descriptive Statistics - Agriculture and Labor Markets in India (2001–2007)

	logAgricultural Outcomes				
	Yield (All Crops)	Value (All Crops)	Price (All Crops)		
DAILY AVERAGE TEMPERATURE (°C)	-0.127^{***} (0.0358)	-0.126^{***} (0.0323)	0.000957 (0.0107)		
Monsoon Rainfall (100mm)	$\begin{array}{c} 0.0115^{***} \\ (0.00375) \end{array}$	0.0107^{***} (0.00347)	-0.000795 (0.00181)		
Fixed Effects	Crop \times District, Crop \times Year and State-Year Time Trends				
Observations	9,813	9,813	9,813		

Table 2: The Effects of Weather on Agricultural Outcomes

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

	log Average Day Wages					
	Agriculture	Manufacturing	Services	Construction		
Daily Average Temperature (°C)	-0.0516^{**} (0.0240)	-0.0344 (0.0558)	-0.0107 (0.0630)	-0.0150 (0.0452)		
Monsoon Rainfall (100mm)	-0.00611^{*} (0.00365)	-0.0136 (0.00873)	-0.00344 (0.00727)	-0.00171 (0.00700)		
Fixed Effects	DISTRICT, YEAR, STATE-YEAR TIME TRENDS					
Observations	1,067	1,068	1,099	1,035		

	Table 3:	The	Effects	of	Weather	on	Average	Wages
--	----------	-----	---------	----	---------	----	---------	-------

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level. Differences in observations across sectors arise due to missing wage data.

	District Labor Force Shares						
	Agriculture	Manufacturing	SERVICES	Construction	Unemployment		
Daily Average Temperature (°C)	-0.110^{***} (0.0190)	0.0761^{***} (0.0104)	0.0386^{***} (0.00758)	0.00115 (0.00711)	-0.00596 (0.00460)		
Monsoon Rainfall (100 mm)	-0.00298 (0.00309)	0.00297^{*} (0.00161)	0.00102 (0.000976)	-0.00185 (0.00123)	0.000836 (0.000742)		
Fixed Effects	DISTRICT, YEAR, STATE-YEAR TIME TRENDS						
Average Share Observations	$0.445 \\ 1,105$	$0.225 \\ 1,105$	$0.183 \\ 1,105$	$0.066 \\ 1,105$	0.078 1,105		

Table 4: The Effects of Weather on the District Labor Force Share of Employment - By Sector

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

Table 5: The Effects of Weather in Foreign Districts on the Share of Employment in Destination Districts - By Sector

	DESTINATION DISTRICT LABOR FORCE SHARES					
	Agriculture	Manufacturing	Construction	Services	Unemployment	
Local Daily Average	-0.100***	0.0781^{***}	0.0364^{***}	-0.00132	-0.0127^{**}	
Temperature (°C)	(0.0170)	(0.0113)	(0.00763)	(0.00701)	(0.00569)	
Local Monsoon Rainfall	-0.00333	0.00304	0.00161	-0.00181	0.000479	
(100 mm)	(0.00327)	(0.00185)	(0.000980)	(0.00125)	(0.000838)	
Foreign Daily Average	-0.0800	-0.0166	0.0195	0.0209	0.0562^{**}	
Temperature (°C)	(0.0534)	(0.0272)	(0.0329)	(0.0182)	(0.0221)	
Foreign Monsoon Rainfall	0.00172	-0.000899	-0.00491	-0.00000550	0.00410	
(100 mm)	(0.00702)	(0.00408)	(0.00352)	(0.00245)	(0.00308)	
Fixed Effects	DISTRICT, YEAR, STATE-YEAR TIME TRENDS					
Average Share	0.445	0.225	0.183	0.066	0.078	
Observations	1,105	1,105	1,105	1,105	$1,\!105$	

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

	India Wide	WITHIN DISTRICT	Within District Skill Adjusted
Average Wage Gap (Casual Manufacturing Workers)	1.335	1.139	1.094
AVERAGE WAGE GAP	2.210	1.936	1.540
(permanent Manufacturing Workers)			
AVERAGE DAY WAGE IN AGRICULTURE (Rs.)	52.27	52.27	52.27
Year Fixed Effects	Yes	Yes	Yes
District Fixed Effects	No	Yes	Yes
Individual Controls	No	No	YES
Observations	50,832	50,832	50,818

Table 6: Average Wage Gap (Agriculture vs. Manufacturing)

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Individual level controls include age, education, and gender. Estimates are based on individual-level mincerian wage regressions on the working-age population (14-65) controlling for a sector dummy (β) specifying whether the individual is engaged in agricultural, casual manufacturing labor, or permanent manufacturing employment. The wage gap is calculated as $\exp(\beta)$. Individual controls include level of education, age, and gender.





Notes: Share of Contract Workers is defined as the share of contract workers in the labor force conditional on hiring one contract worker, i.e., conditional on hiring contract workers, we observe that Pro-Employer States hire more. Share of Firms is defined as the share of firms that hire any contract workers, i.e., Firms in Pro-Worker States are more likely to hire at least one contract worker compared to Pro-Employer States.

	Rigid States	Regulated Firms Flexible States	Difference	RIGID STATES	Unegulated Firms Flexible States	Difference
Total Output	2,130.215	1,572.613	-557.602	132.982	89.374	-43.607
(Million Rs.)	(609.347)	(508.357)	(869.733)	(37.141)	(16.622)	(46.319)
Total Employment	415.447	477.500	62.053	34.702	39.957	5.255
(Non-mangers)	(94.964)	(113.593)	(198.128)	(7.047)	(4.833)	(7.563)
Employment (Contract Workers)	54.354 (11.793)	191.157 (74.758)	$136.802 \\ (75.682)$	$12.361 \\ (4.320)$	9.977 (2.785)	-2.384 (5.716)
Average Day Wage	156.273	140.061	-16.211	145.138	$112.971 \\ (8.846)$	-32.167
(Contract Workers)	(11.194)	(7.689)	(17.130)	(29.158)		(36.825)
Employment (Regular Workers)	361.076 (54.977)	$286.332 \\ (25.177)$	-74.744 (70.039)	$22.340 \\ (4.170)$	29.980 (3.719)	7.640 (5.655)
Average Day Wage	302.003	200.478	-101.525	$195.141 \\ (47.432)$	129.019	-66.121
(Regular Workers)	(71.131)	(23.244)	(87.473)		(12.269)	(58.600)
Employment	53.119	42.526	-10.593	6.827	4.605	-2.221
(Managers)	(11.282)	(7.135)	(16.136)	(2.086)	(0.677)	(2.547)
Average Day Wage	871.471	658.062	-213.409	646.620	371.388	-275.232
(Managers)	(173.962)	(65.288)	(219.641)	(137.602)	(41.575)	(171.595)
Capital	$\substack{1,483.548\\(453.552)}$	1,229.230	-254.318	80.357	50.814	-29.543
(Million Rs.)		(407.124)	(640.261)	(27.135)	(15.276)	(32.173)
ITEMS PRODUCED	2.997	2.582	-0.414	2.482	2.067	-0.414
	(0.558)	(0.206)	(0.700)	(0.336)	(0.170)	(0.473)
Access to Electricity $(\%)$	$0.991 \\ (0.005)$	$0.987 \\ (0.004)$	-0.004 (0.007)	$0.973 \\ (0.023)$	$0.965 \\ (0.014)$	-0.007 (0.032)
Generates Own Electricity (%)	0.441 (0.042)	0.643 (0.043)	0.201^{***} (0.060)	$0.142 \\ (0.091)$	$0.398 \\ (0.057)$	0.255^{**} (0.106)
Output per Worker (Million Rs.)	$3.739 \\ (0.941)$	$2.358 \\ (0.523)$	-1.381 (1.289)	$3.047 \\ (0.560)$	1.869 (0.230)	-1.178 (0.718)
$\begin{array}{c} \text{TFPR} \\ \text{(Log)} \end{array}$	6.119	6.082	-0.037	5.468	5.423	-0.045
	(0.058)	(0.066)	(0.110)	(0.029)	(0.061)	(0.083)

Table 7: Descriptive Statistics - Manufacturing Firms in India (2001–2007)

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Rigid States = 0, Flexible States = 1. The sample is restricted to regulated firms. Standard errors are clustered at the State Level.

	log Output and Employment					
	TOTAL OUTPUT	Items Produced	Employment (All)	Employment (Contract)	Employment (Permanent)	
Panel A: Net Effect						
Daily Average Temperature (°C)	0.0888^{**} (0.0346)	-0.00753 (0.0106)	$\begin{array}{c} 0.0324^{**} \\ (0.0149) \end{array}$	-0.0445 (0.0304)	0.0326^{*} (0.0171)	
Panel B: Differential Effect						
Daily Average Temperature (°C)	0.119^{*} (0.0708)	-0.0282 (0.0271)	0.147^{***} (0.0289)	$0.0830 \\ (0.0887)$	0.0959^{**} (0.0407)	
Temperature × Flexible	-0.0470 (0.0992)	$\begin{array}{c} 0.0321 \\ (0.0384) \end{array}$	-0.177^{***} (0.0436)	-0.201 (0.136)	-0.0985 (0.0608)	
RAINFALL CONTROLS	Yes	Yes	Yes	Yes	Yes	
Fixed Effects	Sector \times District, Sector \times Year, and State-Year Time Trends					
Observations	65,934	65,934	65,921	21,751	60,302	

Table 8: The Effects of Temperature on Manufacturing Firms – Unregulated Firms

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Panel A presents the results from regression the outcome variable on the level of temperature and monsoon rainfall. Panel B presents the differential effects of temperature across labor regulation environments. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

Table 9:	The Effects	of Temperature or	n Manufacturing Firms –	Unregulated Firms
		1	0	0

	log Wages and Productivity						
	Avg. Day Wage (All)	Avg. Day Wage (Contract)	Avg. Day Wage (Permanent)	Output per Worker	TFPR		
Panel A: Net Effect							
Daily Average Temperature (°C)	0.0110 (0.00898)	0.00440 (0.0129)	0.0181^{*} (0.0108)	0.0449 (0.0300)	$\begin{array}{c} 0.0149\\ (0.0156) \end{array}$		
Panel B: Differential Effect							
Daily Average Temperature (°C)	-0.0236 (0.0275)	-0.0171 (0.0329)	-0.00817 (0.0306)	-0.0372 (0.0684)	$0.0255 \\ (0.0388)$		
Temperature × Flexible	$0.0536 \\ (0.0388)$	$0.0340 \\ (0.0517)$	0.0409 (0.0422)	0.127 (0.0973)	-0.0167 (0.0537)		
RAINFALL CONTROLS	Yes	Yes	Yes	Yes	Yes		
Fixed Effects	Sector \times I	District, Sector \times	YEAR, AND STATE-	Year Time Trei	NDS		
Observations	65,921	21,751	60,302	65,934	57,143		

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Panel A presents the results from regression the outcome variable on the level of temperature and monsoon rainfall. Panel B presents the differential effects of temperature across labor regulation environments. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

	log Output and Employment					
	TOTAL OUTPUT	Items Produced	Employment (Contract)	Employment (Permanent)		
Panel A: Net Effect						
Daily Average Temperature (°C)	-0.0101 (0.0341)	-0.0516 (0.0333)	$0.0383 \\ (0.0236)$	-0.0163 (0.0114)		
Panel B: Differential Effect						
Daily Average Temperature (°C)	-0.0971^{*} (0.0540)	-0.0497^{**} (0.0220)	-0.170^{**} (0.0667)	0.00283 (0.0382)		
Temperature × Flexible	0.153^{*} (0.0880)	0.0588^{*} (0.0337)	0.211^{**} (0.104)	$0.0626 \\ (0.0505)$		
RAINFALL CONTROLS	Yes	Yes	Yes	Yes		
Fixed Effects	Sector \times Distr	ict, Sector >	× Year, and Stati	e-Year Time Trends		
Observations	36,985	36,985	18,712	35,818		

Table 10: The Effects of Temperature on Manufacturing Firms – Regulated Firms

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Panel A presents the results from regression the outcome variable on the level of temperature and monsoon rainfall. Panel B presents the differential effects of temperature across labor regulation environments. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

Table 11	l: The	Effects	of Tem	perature	on	Manufactu	ring	Firms -	Regu	ilated	Firms
									()-		

	log WAGES AND PRODUCTIVITY						
	Avg. Day Wage (Contract)	Avg. Day Wage (Permanent)	Output per Worker	TFPR	$\begin{array}{c} \mathrm{TFPR} \\ \mathrm{(CES)} \end{array}$		
Panel A: Net Effect							
Daily Average Temperature (°C)	-0.0299 (0.0299)	-0.0413^{**} (0.0178)	-0.0434^{**} (0.0184)	-0.0301^{*} (0.0159)	-0.0299^{***} (0.0110)		
Panel B: Differential Effect							
Daily Average Temperature (°C)	$0.0335 \\ (0.0333)$	-0.0774^{***} (0.0221)	-0.103^{**} (0.0513)	-0.0958^{***} (0.0299)	-0.100^{***} (0.0313)		
Temperature × Flexible	-0.113^{**} (0.0495)	0.0838^{**} (0.0352)	0.128 (0.0838)	0.0969^{**} (0.0438)	0.101^{**} (0.0446)		
RAINFALL CONTROLS	Yes	Yes	Yes	Yes	Yes		
Fixed Effects	Sector \times Dis	STRICT, SECTOR \times Y	EAR, AND STATE	-Year Time	Trends		
Observations	18,712	35,818	36,985	33,440	33,464		

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Panel A presents the results from regression the outcome variable on the level of temperature and monsoon rainfall. Panel B presents the differential effects of temperature across labor regulation environments. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

	log Capital, Management, and Entry						
	Capital	CAPITAL DEPRECIATION	Employment Managers	Day Wage Managers	Number of Plants		
Panel A: Net Effect							
Daily Average Temperature (°C)	$\begin{array}{c} 0.0571 \\ (0.0442) \end{array}$	$0.0329 \\ (0.0263)$	0.0323 (0.0220)	-0.0240^{*} (0.0144)	0.00677 (0.00646)		
Panel B: Differential Effect							
Daily Average Temperature (°C)	$0.104 \\ (0.0730)$	$0.0595 \\ (0.0530)$	-0.00464 (0.0442)	-0.0358 (0.0315)	0.00179 (0.0156)		
Temperature × Flexible	-0.0830 (0.114)	-0.0482 (0.0823)	0.0651 (0.0718)	0.0208 (0.0496)	0.00879 (0.0242)		
RAINFALL CONTROLS	Yes	Yes	Yes	Yes	Yes		
Fixed Effects	Sector \times	DISTRICT, SECTO	$DR \times YEAR$, AND	STATE-YEAR	Time Trends		
Observations	36,810	$29,\!454$	$36,\!550$	36,550	36,985		

Table 12: Additional Manufacturing Firm Outcomes – Regulated Firms

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

	Total GDP	Agricultural GDP	Services GDP	Manufacturing GDP	Construction GDP
Daily Average Temperature °C	-0.0263^{*} (0.0136)	-0.116^{**} (0.0571)	-0.0257^{*} (0.0144)	-0.0120 (0.00963)	0.0183 (0.0232)
RAINFALL CONTROLS	Yes	Yes	Yes	Yes	Yes
Fixed Effects		DISTRICT,	Year, and State	e-Year Time Trends	
Observations	3,432	3,432	3,432	3,432	3,432

Table 13: The Effects of Temperature on GDP

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. All dependent variables are in logs. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

	Baseline	Shutting Down Labor Reallocation		TEMP	No advers erature Ei	E FFECTS
	(1)	(2)	(3)	(4)	5	6
Informal (34%)	-18.3%	-18.3%	-18.3%	-18.3%	-18.3%	-3%
UNREGULATED FORMAL (44%)	8.8%	8.8%	-6.5%	8.8	24.1%	24.1%
Regulated Formal (22%)	-1.01%	$-15.3\% \times {\rm Flexibility}$	$-15.3\% \times \text{Flexibility}$	15.3%	15.3%	15.3%
Total Manufacturing Effect	-2.57%	-4.58%	-11.32%	1.01%	7.74%	12.95%
Total Effect (Aggregate)	-3.03%	-3.31%	-4.25%	-2.53%	-1.59%	-0.86%
Change (%)	_	9.24%	40.26%	-16.50%	-47.52%	-71.67%

Table 14: Counterfactual Estimates

NOTES: Column 1 (Baseline) provides a decomposition of the effect of a 1°C increase in temperature on manufacturing GDP decomposed into the regulated and unregulated formal manufacturing sector, using the estimated effects of a 1°C increase in temperature on firm-level output, and the informal sector, whereby the effect is the residual effect of a 1°C increase in temperature on firms in the informal sector to produce the estimated effect on manufacturing GDP. Columns 2, and 3 consider the effects of increasing the rigidity of the labor market environment to the level of West Bengal (the most Pro-Worker State). Column 2 increases the rigidity within the regulated formal manufacturing sector, and column 3 increases the rigidity in both the regulated formal manufacturing sector, equivalent to expanding the coverage of the IDA. Columns 4, 5 and 6 consider the consequences of shutting down the adverse effects of temperature on manufacturing. Column 4 turns off the adverse effects of temperature for firms in the regulated formal manufacturing sector. Column 5 turns off the adverse effects of temperature for firms in the regulated formal manufacturing sector (under the assumption that the adverse effects are constant across these two sectors). Column 6 turns off the adverse effects of temperature for firms in the formal and informal sector (under the assumption that the adverse effects are constant across these three sub-sectors).

Online Appendices – Not for Publication

A Theory Appendix

This appendix presents a simple model based on Matsuyama (1992) demonstrating how the direction of labor reallocation in response to a sector-specific productivity shock depends on market integration. Any analysis of labor reallocation across sectors within an economy necessitates a diversified economy and so for simplicity I consider two sectors: agriculture (a) and manufacturing (m).

Preferences

Consider a country composed of a large number of regions *i*. Each location *i* is populated by a continuum of workers L_i , which are assumed to be mobile between sectors, immobile between regions, supplied inelastically, and fully employed. Workers earn income $w_{ij}L_{ij}$ and preferences are defined over two types of goods agriculture and manufactured goods. Agricultural consumption is subject to subsistence constraints with a Stone-Geary utility function (Matsuyama, 1992; Caselli and Coleman, 2001; Jayachandran, 2006; Desmet and Parente, 2012).²⁸ Given prices in sector *j*, p_{ij} and total income w_iL_i , each worker maximises

$$U_i = (C_{ia} - \bar{a})^{\alpha} C_{im}^{1-\alpha} \tag{3}$$

which they maximise subject to their budget constraint,

$$p_{ia}C_{ia} + p_{im}C_{im} \le L_i w_i \tag{4}$$

Worker demand for goods in agriculture, $D_{ia} = p_{ia}\bar{a} + \alpha(L_iw_i - p_{ia}\bar{a})$. For manufactured goods $D_{im} = (1 - \alpha)(L_iw_i - p_{ia}\bar{a})$. As such, preferences are non-homothetic. Higher food subsistence requirements, higher prices, and lower incomes are associated with an increase in the demand for agricultural goods (D_{ia}/L_iw_i) .

Production

There are 2 goods that can be produced in each location i, agricultural good a and manufactured goods m.²⁹ I assume that all regions have access to the same technology and so

²⁸Non-homothetic preferences can also be incorporated through a CES utility function where the elasticity of substitution between agricultural goods and other goods is less than one (Ngai and Pissarides, 2007; Desmet and Rossi-Hansberg, 2014).

²⁹I will refer to goods and sectors interchangeably.

production functions do not differ across regions within each industry. Different industries may have different production functions. Consequently, I drop the locational subscript unless necessary.

Output of each good j is produced according to the following production function,

$$Y_j = A_j F_j(L_j) \tag{5}$$

where A_j is sector-specific productivity and L_j is the set of workers in sector j. I assume that $F_j(0) = 0$, $F'_j > 0$ and $F''_j < 0$. In addition, I assume that $A_a F^1(1) > \bar{a}L > 0$. This inequality states that agriculture is productive enough to provide the subsistence level of food to all workers. If this condition is violated then workers receive negative infinite utility.

Each firm equates its demand for labor to the value of the marginal product of labor. Consequently, as market clearing requires that $L_a + L_m = L$, the marginal productivity of labor will be equalised across sectors,

$$p_a A_a F_a'(L_a) = w = p_m A_m F_m'(L_m) \tag{6}$$

Equilibrium

Autarky and Equilibrium Prices

Equilibrium is defined as a set of prices, wages, and an allocation of workers across sectors such that goods and labor markets clear. In a state of autarky, the price ensures that the total amount produced is equal to total consumption in each location, so that,

$$C_a = A_a F_a(L_a) \tag{7}$$
$$C_m = A_m F_m(L_m)$$

Maximisation of equation 3 implies that each worker consumes agricultural goods such that,

$$p_a C_a = \bar{a} + \frac{\alpha p_m C_m}{1 - \alpha} \tag{8}$$

Combining this result with the profit maximisation condition (equation 6), the labor market clearing condition $(L_m = 1 - L_a)$, and the fact that total production must equal total consumption yields,

$$\Omega(L_m) = \frac{\bar{a}}{A_a} \tag{9}$$

where,

$$\Omega(L_m) \equiv F_m(L_m) - \frac{F_m'(L_m)F_a(1-L_a)}{F_a'(1-L_a)}$$
(10)

In addition, it is the case that $\Omega(0) = F_m(1)$, $\Omega(1) < 0$ and $\Omega'(\cdot) < 0$.

Consequently, in equilibrium a unique interior solution will arise for the employment share in manufacturing L_m ,

$$L_m = \Omega^{-1} \left(\frac{\bar{a}}{A_a} \right) \tag{11}$$

As preferences are non-homothetic the demand for agricultural goods (food) decreases as income increases (Engel's law). Consequently, an increase (decrease) in agricultural productivity will push (pull) workers into the manufacturing (agricultural) sector. Similarly, a decrease (increase) in the subsistence constraint \bar{a} will push (pull) workers into the manufacturing (agricultural) sector.

Trade and Equilibrium Prices

Without opportunities to trade, consumers must consume even their worst productivity draws. The ability to trade breaks the production-consumption link. In the case of free trade prices, set globally, are taken as given. If the world price for a good j, \bar{p}_j , exceeds the autarkic local price p_{ij} , firms and farms will engage in arbitrage and sell to the global market. By contrast, if the world price for a good j is less than the autarkic local price consumers will import the product from outside of the local market. Consequently, local demand does not affect the allocation of labor across sectors, i.e., changes in A_{ij} do not affect prices.

As discussed above the rest of the world differs only in terms of agricultural and manufacturing productivity, $A_{i'a}$ and $A_{i'm}$. Profit maximisation in the rest of the world implies that,

$$p_a A_{i'a} F_{i'a}'(L_{i'a}) = p_m A_{i'm} F_{i'm}'(L_{i'm})$$
(12)

Within industry production functions are assumed to be constant across regions. Under the assumption of free trade and incomplete specialisation manufacturing employment in region i, L_{im} , is now determined jointly by equations 6 and 12. Taking the ratio of these equations provides the following equality,

$$\frac{F_{im}'(L_{im})}{F_{ia}'(L_{ia})} = \frac{A_{ia}A_{i'm}}{A_{i'a}A_{im}} \frac{F_{i'm}'(L_{i'm})}{F_{i'a}'(L_{i'a})}$$
(13)

As $\frac{F_{im}'(L_{im})}{F_{ia}'(L_{ia})}$ is decreasing in L_{im} it follows that,

$$L_{im} \gtrless L_{ia} \quad iff \quad \frac{A_{i'a}}{A_{i'm}} \gtrless \frac{A_{ia}}{A_{im}}$$

$$\tag{14}$$

In this case an increase (decrease) in agricultural productivity will pull (push) workers into the agricultural (manufacturing) sector, due to a change in local comparative advantage. This is demonstrated in figure 1

Figure 1: The Effect of a Reduction in Agricultural Productivity on Equilibrium Employment Shares (Free Trade)



In the case of costly trade, firms (farms) will engage in aribtrage opportunities as before; however, the local price is bounded by a trade cost δ . Consequently, a trader will engage in arbitrage, selling on the global market, as long as the global price is greater than the local price net of trade costs, i.e., $\bar{p}_j/\delta > p_j^A$. Conversely, consumers will import from the global market if the local price is greater than the global price net of trade costs, i.e., $\bar{p}_j < p_j^A/\delta$. Consequently, in the case of homogenous traders where all agents face a constant iceberg trade cost, the local price is bounded by the global price, i.e., $\frac{\bar{p}_j}{\delta} \leq p_j^A \leq \bar{p}_j \delta$.

B Data appendix

B.1 Agricultural Data Appendix

This section provides additional details on the Agriculture data used in section III.

As discussed in the main paper, the data is collected from the ICRISAT Village Dynamics in South Asia Macro-Meso Database (henceforth VDSA) which is compiled from a number of official government datasources. Figures 1 provides summary statistics for the 12 crops used.



Figure 1: Average: (TL) Production; (TR) Cultivated Land Area; (BL) Yield; (BR) Price (2001 Rs.)

We observe from the figures that both Rice and Wheat are the most produced crops in terms of cultivated land area and total production (figure 1) and that they also comprise the largest share of production and cultivated land area within-district (figure 2). However, in terms of yields sugarcane is show to have one of the highest yields and has the largest share of yields within-district (figure 2).



Figure 2: District Shares of: (TL) Agricultural Production; (TR) Agricultural Cultivated Land; (B) Agricultural Yields

B.2 NSS Data Appendix

This section provides additional details on the NSS Employment and Unemployment surveys used in section III. The National Sample Survey Organisation (NSSO) carries out all-India, large sample, household surveys on employment and unemployment every few years. This paper takes advantage of the 60th round (January 2004 – June 2004), the 61st round (July 2004 – June 2005), the 62nd round (July 2005 – June 2006), and the 64th round (July 2007 – June 2008).

Using this data I construct average day wage for agricultural workers, manufacturing workers, services workers and construction workers. Looking at the breakdown of employment between rural and urban areas it is clear that non-agricultural activities are not restricted to urban areas.

As one might expect agricultural employment is largely focused in rural areas accounting for an average of 60% of rural employment during this period. However, employment manufacturing and services account collectively for just over 25% of rural employment. By

	Rural	Urban	Combined
Agriculture	59.7%	7.8%	42.7%
Manufacturing	14.6%	40%	22.9%
Services	11.5%	35.7%	19.4%
Construction	5.9%	7.6%	6.5%
Unemployment	8.1%	8.6%	8.3%

Table B1: Labor Force Shares in India (2001–2007)

contrast, in urban areas manufacturing and services account for close to 75% of employment. This is consistent with one of the most striking features of India's recent spatial development, namely the expansion of India's metropolitan areas into rural areas, referred to peri-urbanization (see Colmer (2015) for a more detailed discussion and review of this literature). In the last decade there has been an official increase in urban agglomerations by 25% with populations shifting outwards. Henderson (2010) presents evidence in support of this industrial decentralization for the Republic of Korea and Japan. Desmet et al. (forthcoming) and Ghani et al. (2014) also provide supporting evidence for this process in India. Desmet et al. (forthcoming) show that the services sector has become increasingly concentrated over time, while manufacturing has become less concentrated in districts that were already concentrated and has increased in districts which originally were less concentrated. Ghani et al. (2014) look more specifically at the manufacturing sector and document its movement away from urban to rural areas, comparing the formal and informal sectors. The authors argue that the formal sectors is becoming more rural; however, in practice a lot of this movement is likely sub-urbanization, rather than ruralisation, in which firms move to the outskirts of urban areas where they can exploit vastly cheaper land and somewhat cheaper labor. Colmer (2015) finds evidence consistent with these papers finding that manufacturing employment growth has become more concentrated in districts which were initially less concentrated, and that this employment growth is significantly higher in less concentrated rural areas compared to less concentrated urban areas.

This process of peri-urbanization also benefits workers reducing the cost of sectoral adjustment and migration costs. Indeed, in many instances it may reduce the need to migrate altogether with workers choosing to commute from home, rather than migrate to urban areas. This is consistent with the non-trivial shares of manufacturing employment and agricultural employment presented in rural and urban areas respectively. Interestingly, we observe that the unemployment share in urban areas is almost twice the size of those in rural areas, suggesting that there is more absorptive capacity in rural areas.

B.3 Weather Data Appendix

This section provides additional details on the weather data used throughout this paper.



Figure 3: Intra-Annual Weather Variation



Figure 4: Inter-Annual Weather Variation



Figure 5: Spatial Weather Variation

B.4 ASI Data Appendix

This section provides additional details on the Annual Survey of Industries Establishmentlevel Microdata.

I begin by extracting a subset of variables from the raw data separately for each year and then append each year together before apply the following cleaning processes, summarised in table B2 With this initial sample I begin by dropping all plants that are outside of the manufacturing sector, closed. In addition, I remove all observations with missing or zero total output data due to the importance of the revenue and productivity results. I then combine this data with the weather data taken from the ERA-Interim Reanalysis Data archive. Finally, I drop Union Territories and remove all districts with zero agricultural production. This is due to the focus on agricultural productivity shocks as a driver of labor reallocation.

All financial amounts are deflated to constant 2001-02 Rupees.³⁰ Revenue (gross sales) is deflated by a three-digit commodity price deflator available from the "Index Numbers of Wholesale Prices in India - By Groups and Sub-Groups (Yearly Averages)" produced by the Office of the Economic Adviser in the Ministry of Commerce & Industry.³¹ Material inputs are deflated by constructing the average output deflator for a given industry's supplier industries based on India's 1993-94 input-output table, available from the Central Statistical Organization. Fuel and Electricity costs are deflated by the price index for "Fuel, Power, Light, and Lubricants". Capital is deflated by an implied national deflator calculated from "Table 13: Sector-wise Gross Capital Formation" from the Reserve Bank of India's Handbook of Statistics on the India Economy.³² Wage costs are deflated using a national GDP deflator.

 $^{^{30}{\}rm Thank}$ you to Hunt Allcott, Allan Collard-Wexler, and Stephen O'Connel for publicly providing the data and code to conduct this exercise.

³¹Available from http://www.eaindustry.nic.in/

 $^{^{32}\}mathrm{Available}$ from http://www.rbi.org.in

Action Taken	Observations Dropped	Final Sample
Initial Sample	-	371,383
Drop Sectors Outside of Manufacturing	22,645	348,739
Drop Closed Plants	90,115	258,624
Merge Weather Data	$16,\!196$	$235,\!679$
Drop Union Territories	1,727	$233,\!952$
Drop if Employment < 10	56,269	177,683
Drop if Employment < 20 & No Electricity	1,139	176,544
Drop Total Output Zero or Missing	19,989	$156,\!555$
Merge with Deflators	9	156,546
KEEP AGRICULTURAL DISTRICTS	53,273	103,273
Plants-Year Observations Above the Threshold	-	36,985
Plants-Year Observations Below the Threshold	-	$65,\!934$

 Table B2: ASI Sample Selection

C Productivity Estimation

In what follows I provide an explicit model of TFPR, in the context of a profit-maximising firm.

Each firm *i*, in time *t*, produces output Q_{it} using the following (industry-specific) technology:

$$Q_{it} = A_{it} K_{it}^{\alpha_k} M_{it}^{\alpha_m} E_{it}^{\alpha_E} L_{it}^{\alpha_L}$$

where K_{it} is the capital input, L_{it} is the labor input, M_{it} is the materials input, and E_{it} is the electricity input. Furthermore, I assume constant returns to scale in production so $\alpha_M + \alpha_E + \alpha_K + \alpha_L = 1$.

The demand curve for the firm's product has a constant elasticity:

$$Q_{it} = B_{it} P_{it}^{-\epsilon}$$

Combining these two equations I obtain an expression for the sales-generating production function:

$$S_{it} = \Omega_{it} K_{it}^{\beta_k} M_{it}^{\beta_M} E_{it}^{\beta_E} L_{it}^{\beta_I}$$

where $\Omega_{it}(true) = A_{it}^{1-\frac{1}{\epsilon}} B_{it}^{\frac{1}{\epsilon}}$, and $\beta_X = \alpha_X (1-\frac{1}{\epsilon})$ for $X \in \{K, L, M, E\}$. Within the confines of this paper, I define true productivity as $\omega_{it} \equiv \log(\Omega_{it})$.

To recover a measure of ω_{it} , I compute the value of β_L , β_M , and β_E using median regression for each industry-year cell.

$$\beta_X = median\left(\left\{\frac{P_{it}^X X_{it}}{S_{it}}\right\}\right) \quad \text{for } X \in \{L, M, E\}$$

To recover the coefficient on capital, β_K , I use the assumption of constant returns to scale in production, i.e., $\sum_X \alpha_X = 1$, such that:

$$\beta_K = \frac{\epsilon - 1}{\epsilon} - \beta_L - \beta_M - \beta_E$$

For ease of measurement I set ϵ to be constant for all firms. Following Bloom (2009) I set $\epsilon = 4$. Using these estimates I compute ω_{it} ,

$$\omega_{it}(est) = \log(S_{it}) - \beta_K \log(K_{it}) - \beta_M \log(M_{it}) - \beta_E \log(E_{it}) - \beta_L \log(L_{it})$$

C.1 Allowing for Differences in the Elasticity of Substitution Within Labor

As suggested by the empirical results contract labor does not appear to perfectly substitutable with permanent labor as implied under the Cobb-Douglas production function. This section presents an alternative production function, used to estimate productivity allowing for imperfect substitutability between these two labor types. Specifically, I estimate a nested Cobb-Douglas production function, in which the aggregate labor factor is a CES function of Contract and Permanent Labor.

As above the top-level sales-generating production function is Cobb-Douglas,

$$S_{it} = \Omega_{it} K_{it}^{\beta_k} M_{it}^{\beta_M} E_{it}^{\beta_E} L_{it}^{\beta_I}$$

However, the Labor input is CES, i.e.,

$$L_{it} = \left[\theta_c L_{cit}^{\frac{\sigma-1}{\sigma}} + \theta_p L_{pit}^{\frac{\sigma-1}{\sigma}}\right]^{\frac{\sigma}{\sigma-1}}$$

In the event that contract workers and permanent workers are perfectly substitutable this production function collapses back to the standard Cobb-Douglas production function. Given the results presented in the main text, each of the parameters in the CES structure are observed or estimated. $\theta_c L_{cit} = \bar{w}_{cit} L_{cit}$, i.e. the wage bill of the firm for each labor type.

Given that contract and permanent labor markets are segmented, i.e., we observe no increase in the number of permanent workers, this suggests that the tasks that the casual entrants and permanent workers engage in are complementary in production. In light of this, it is possible to provide an exogenous estimate of the elasticity of substitution, σ , between the new entrants into casual positions and the incumbent permanent workers. If $\sigma < 1$ the new entrant casual workers and incumbent permanent workers engage in tasks that are complementary in production. If $\sigma > 1$ then these workers engage in tasks that are substitutable in the production process.

$$\sigma \propto \frac{\partial \log w_m^p}{\partial Temperature} / \frac{\partial \log L_m^c}{\partial Temperature} = \frac{\partial \log w_m^p}{\partial \log L_m^c} = 0.39$$
(15)

These results suggest that a 1% increase in the number of casual workers, employed out of agriculture, is associated with a 0.39% increase in the average wage of permanent manufacturing workers. To the degree that new entrants out of agriculture and incumbent casual workers are substitutable in tasks, this would indicate that, on average, contract and permanent workers in the regulated Indian manufacturing sector engage in complementary production tasks. With these parameters in hand I construct L_{it}^{CES} for each firm and then estimate productivity using the CES labor input in place of the Cobb-Douglas Labor input.

D The Labor Regulation Environment – Supporting Evidence

This appendix provides supporting evidence for the identification strategy that exploits spatial variation and firm-level exposure to India's labor regulation environment.

D.1 Bunching in the Firm-Size Distribution

First I examine the degree to which there is bunching in the firm-size distribution, exploiting differences in the incentives that firms face across different states as well as differences in the regulatory thresholds. Previous work has argued that there little evidence of bunching in the firm-size distribution associated with the Industrial Disputes Act (Hsieh and Olken, 2013). However, I demonstrate that the absence of bunching in this previous work arises due to three considerations: 1) applying a nationwide threshold of 100, rather than state-specific thresholds; 2) not distinguishing between pro-worker (where bunching is more likely) and pro-employer states (where bunching is less likely); 3) conflation between the effect of the IDA and changes in the sampling schedule around the modal threshold of 100 workers. In the results presented below I use the year 2007 as this has the largest sample to maximize power; however, results are robust to other years. Furthermore, all results account for the sampling weights in order not to minimize the conflation of changes in the sample schedule with changes arising from the IDA.

I demonstrate that in West Bengal, arguably the state with the most rigid labor regulation environment, that there is a bunching of firms just below the regulatory threshold of 50 workers. However, there is no bunching for the other states around this threshold, in support of the identification strategy.



Identifying bunching around the regulatory threshold of 100 is more challenging as this coincides with a change in the sampling scheme of the ASI, in which there is an oversampling

above this threshold. Despite accounting for sampling weights, we observe that there is limited bunching just *above* the regulatory threshold. This is the complete opposite to what one should expect. In rigid states there should be bunching to the left of the threshold, and for flexible states this bunching should be smaller, consistent with the observed bunching in West Bengal at the 50 worker threshold. Due to the fact that the sampling scheme of the ASI coincides with the regulatory threshold it is impossible to identify or rule out bunching at this level. However, it is interesting to note that we observe stronger bunching to the right of a placebo regulatory threshold of 100 for West Bengal and Uttar Pradesh, where there is no actual threshold in place. This provides support for the conclusion that the sampling scheme of the ASI interferes with the identification. The limited bunching to the right of the regulatory threshold in the other states, suggests that in the absence of the change in the sampling scheme we would observe some though limited bunching to the left of the regulatory threshold at 100 workers.



In further support of the assumption that the sampling scheme interferes with the identification strategy I don't find any differences around the regulatory threshold of 300 in "flexible" Uttar Pradesh. This suggests that there isn't anything fundamentally related to the labor regulation environment in flexible states that would result in bunching to the right of the regulatory threshold.



Collectively, this evidence suggests that by accounting for state-specific thresholds and distinguishing between flexible and rigid labor markets (as opposed to grouping all states), one can uncover evidence of bunching in the firm-size distribution associated with the Industrial Disputes Act. Consequently, this suggests that regulated firms may well face differential incentives in the propensity to hire workers in response to transitory labor demand shocks.

D.2 Temperature isn't correlated with Amendments made to the Industrial Disputes Act

In this section I demonstrate that temperature didn't appear to be a determinant of amendments made to the IDA, using data on the year and state of amendments made. Given that the weather data is only available from 1979, I am unable to look at amendments prior to this. Between 1979 and 1995 18 amendments out of a total of 39 were made.

The results suggest that temperature is not correlated with the introduction, direction, or magnitude or amendments made. Even the largest coefficient in column (3) accounting for the magnitude of amendment changes, suggests that a 1 standard deviation increase in temperature is associated with at most an additional 0.022 pro-worker amendments. Rainfall does not appear to be correlated with whether an amendment was mode or the direction of amendment changes on average. However, when the magnitude changes are taken into account a one standard deviation increase in rainfall is associated with an additional 0.12

	(1) Any Change	(2) Relative Change	(3) Total Change
Daily Average Temperature (°C)	0.00107 (0.0584)	$0.0310 \\ (0.0530)$	$0.0682 \\ (0.119)$
Monsoon Rainfall (100mm)	$0.0167 \\ (0.0128)$	$0.0176 \\ (0.0109)$	0.0677^{**} (0.0263)
Observations	272	272	272

Table D1: The Effects of Temperature on Amendments to the Industrial Disputes Act

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. The unit of analysis is a state-year. ANY CHANGE relates to whether an amendment was made. RELATIVE CHANGE accounts for the direction of any amendment changes, i.e., whether it was pro-worker or pro-employer. TOTAL CHANGE accounts for the magnitude and direction of the change, e.g., if 3 pro-worker amendments were made a value of -3 would be assigned to that state in that year. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

pro-worker amendments. While statistically significant, the magnitude of this effect is not particularly large. Most importantly, this does not affect the inference associated with the evaluation of temperature.

D.3 The Effect of the Labor Regulation Environment on Unregulated Sectors

Finally, I explore whether there are any noticeable differences in temperature effects across labor regulation environments through an examination of unregulated sectors such as agriculture, services and construction. I find limited evidence to suggest that there are significant differences in the effects temperature effects across labor regulation environments when looking at these unregulated sectors, or the manufacturing broadly defined to include both the formal sector (those above and below the regulatory threshold) and the informal sector. These results provide further support for the identification strategy as they suggest that there are no first-order spatial differences between the collection of states that make up rigid and flexible labor regulation environments, that are likely to bias the estimated effects on regulated firms.

Table D2: The Differential Effect of Temperature on Real GDP - By Sector (2001 – 2012)

	(1) Total	(2) Agriculture	(3) Manufacturing	(4) Services	(5) Construction
Daily Average Temperature (°C)	-0.00221 (0.0145)	-0.0939^{*} (0.0543)	-0.0759^{**} (0.0344)	0.0113 (0.0166)	0.112^{**} (0.0503)
Temperature × Flexibility	-0.0353 (0.0226)	-0.0324 (0.0798)	0.0736 (0.0452)	-0.0342 (0.0236)	-0.137^{*} (0.0709)
RAINFALL CONTROLS	Yes	Yes	Yes	Yes	Yes
Fixed Effects]	District, Yea	r, and State-Yea	ar Time Tr	RENDS
Observations	3,432	3,432	3,432	3,432	3,432

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

E Additional Results and Robustness Tests

This appendix provides a series of additional results and robustness tests to support of the main results presented in the paper.

E.1 Non-Linearities in the Temperature Schedule

In this section I explore the degree to which there are non-linearities in the temperature schedule. A large literature in agricultural science has demonstrated that the relationship between agricultural yields and weather is highly nonlinear (Schlenker and Roberts, 2009; Auffhammer and Schlenker, 2014). To account for these non-linearities I explore two exercises. First, I apply the concept of growing degree days, which measure the amount of time a crop is exposed between a given lower and upper bound with daily exposures summed over the season. Denoting the lower bound as t_l , the upper bound as t_h , and t_d as the daily average temperature on a given day,

$$GDD_{d;t_l;t_h} = \begin{cases} 0 \text{ if } t_d \leq t_l \\ t_d - t_l \text{ if } t_l < t_d < t_h \\ t_h - t_l \text{ if } t_h \leq t_d \end{cases}$$
(16)

These daily measures are then summed over the period of interest.³³ This approach is appealing for several reasons. First, the existing literature suggests that this simple function delivers results that are very similar to those estimated using more complicated functional forms (Schlenker and Roberts, 2009; Burgess et al. 2016; Burke and Emerick, 2015). Secondly, these other functional forms typically feature higher order terms, which in a panel setting means that the unit-specific mean re-enters the estimation, as is the case with using the quadratic functions (McIntosh and Schlenker, 2006). This raises both omitted variable concerns, as identification in the panel models is no longer limited to location-specific variation over time.

Using the notion of GDD, I model weather as a simple piecewise linear function of temperature and precipitation,

³³For example, if we set t_l equal to 0°C and t_h equal to 24°C then a given set of observations $\{-1, 0, 8, 12, 27, 30, 33\}$, would provide $GDD_{dt;0;24} = \{0, 0, 8, 12, 24, 24\}$. Similarly if we wanted to construct a piecewise linear function setting t_l equal to 24 and t_h equal to infinity the second "piece" would provide $CDD_{dt;24;\infty} = \{0, 0, 0, 0, 6, 9\}$. These values are then summed over the period of interest, in this case $CDD_{dt;0;24} = 68$ and $CDD_{dt;24;\infty} = 15$. This approach accounts for any differences in the response to this temperature schedule relative to a different schedule with the same daily average temperature.
$$f(w_{dt}) = \beta_1 GDD_{dt;t_l;t_h} + \beta_2 GDD_{dt;t_h;\infty} + \beta_3 Rain_{dt}$$

$$\tag{17}$$

The lower temperature "piece" is the sum of GDD between the lower bound $t_l = 0$ and kink-point t_h . The upper temperature "piece" has a lower bound of t_h and is unbounded above. The kink-point in the distribution t_h is determined by estimating an agricultural production function, looping over all possible thresholds and selecting the model with the lowest root-mean-square error. This results in a kink-point at 17°C. This kink-point is applied to all results for consistency.

The second approach explores the effects of non-linearities in the temperature schedule captures the distribution of daily temperatures in district d within year t, by counting the number of days that the daily average temperature fell within the jth bin of 10 temperature bins. I estimate separate coefficients for each of the temperature bin regressors, using the modal bin as a reference category to minimize multicollinearity concerns. So as to retain power, I restrict the lowest bin to contain all days that are < 15°C and the highest bin to contain all days that are > 31°C. Each of the bins between are 2°C wide. Using this approach I model weather as a flexible function of temperature and precipitation,

$$f(w_{dt}) = \sum_{j=1}^{10} \beta_j Temp_{dtj} + \beta_3 Rain_{dt}$$
(18)

This approach makes a number of assumptions about the effects of daily temperatures on the outcomes explored, as discussed in Burgess et al. (2016). First, the approach assumes that the impact of daily temperature is determined by the daily mean alone, rather than intra-day variations in temperature. Second, the approach assumes that the impact of a day's average temperature on the outcome of interest is constant within each 2°C interval. Finally, by using the total number of days in each bin in each year, it is assumed that they sequence of relatively hot and cold days is irrelevant for how hot days affect the annual outcomes.

The results of these exercises are presented below for each group of outcome variables.

E.1.1 Agricultural Outcomes

	Agricultural Outcomes				
	Log Value (All Crops)	Log Yield (All Crops)	Log Price (All Crops)		
DEGREE DAYS (10 days) $t_L = 17, t_H = \infty$	-0.00757^{***} (0.00206)	-0.00728*** (0.00180)	0.000289 (0.000579)		
DEGREE DAYS (10 days) $t_L = 0, t_H = 17$	0.00394 (0.00369)	-0.00437 (0.00475)	-0.00831^{**} (0.00395)		
RAINFALL CONTROLS	Yes	Yes	Yes		
Fixed Effects	Crop \times District, Crop \times Year and State-Year Time Trends				
Observations	9,813	9,813	9,813		

Table E1: The Effects of Daily Temperature on Agricultural Outcomes

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence (up to 1,000km) as modelled in Conley (1999) and serial correlation (1-year) as modelled in Newey and West (1987). District distances are computed from district centroids. Kernels are selected to provide the most conservative standard errors, looped over all distances between 10 and 2,000km and 1-7 years. Results are also robust to using cluster robust standard errors at the State Level.

Figure 1: The Effects of Daily Temperature on Agricultural Outcomes



E.1.2 Employment and Wage Outcomes

	(1) Agriculture	(2) Manufacturing	(3) Services	(4) Construction		
DEGREE DAYS (10 days) $t_L = 17, t_H = \infty$	-0.000918 (0.000734)	-0.00156 (0.00114)	-0.000136 (0.00201)	-0.000518 (0.00119)		
DEGREE DAYS (10 days) $t_L = 0, t_H = 17$	0.000373 (0.00258)	0.00122 (0.00336)	-0.000593 (0.00357)	0.00118 (0.00222)		
RAINFALL CONTROLS	Yes	YES	Yes	Yes		
Fixed Effects	DISTRICT, YEAR AND STATE-YEAR TIME TRENDS					
Observations	1,067	1068	1099	1035		

Table E2: The Effects of Daily Temperature on Wages

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence (up to 1,000km) as modelled in Conley (1999) and serial correlation (1-year) as modelled in Newey and West (1987). District distances are computed from district centroids. Kernels are selected to provide the most conservative standard errors, looped over all distances between 10 and 2,000km and 1-7 years. Results are also robust to using cluster robust standard errors at the State Level.

Figure 2: The Effects of Daily Temperature on Wages



	(1) Agriculture	(2) Manufacturing	(3) Services	(4) Construction	(5) Unemployed
DEGREE DAYS (10 days) $t_L = 17, t_H = \infty$ DEGREE DAYS (10 days) $t_L = 0, t_H = 17$	-0.00264*** (0.000648) -0.00435*** (0.00134)	0.00163*** (0.000436) 0.00269*** (0.000657)	$\begin{array}{c} 0.000796^{***} \\ (0.000259) \\ 0.00169^{**} \\ (0.000759) \end{array}$	0.000107 (0.000183) 0.0000111 (0.000355)	0.000114 (0.000145) -0.0000398 (0.000359)
Rainfall Controls Fixed Effects	Yes	Yes District, Year, A	Yes and State-Yea	Yes r Time Trends	Yes
Observations	1,105	1,105	1,105	1,105	1,105

Table E3: The Effects of Daily Temperature on Employment

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence (up to 1,000km) as modelled in Conley (1999) and serial correlation (1-year) as modelled in Newey and West (1987). District distances are computed from district centroids. Kernels are selected to provide the most conservative standard errors, looped over all distances between 10 and 2,000km and 1-7 years. Results are also robust to using cluster robust standard errors at the State Level.

Figure 3: The Effects of Daily Temperature on Employment



E.1.3 Manufacturing Firms

	Total Output	Employment Contract	Employment Permanent	Day Wage Contract	Day Wage Permanent
Degree Days (10 days)	-0.00250	-0.00396*	0.000167	0.00120	-0.00226***
$t_L = 17, t_H = \infty$	(0.00162)	(0.00204)	(0.00110)	(0.000992)	(0.000621)
DD High \times Flexibility	0.00429^{*}	0.00556^{*}	0.00222	-0.00331**	0.00221^{**}
	(0.00244)	(0.00302)	(0.00142)	(0.00141)	(0.000994)
Degree Days (10 days)	-0.00546	-0.0114	-0.00611	-0.00663	-0.000463
$t_L = 0, t_H = 17$	(0.00730)	(0.0134)	(0.00514)	(0.00540)	(0.00308)
DD Low \times Flexibility	0.00584	0.00990	0.00673	0.00810	0.00221
	(0.0114)	(0.0193)	(0.00780)	(0.00793)	(0.00448)
RAINFALL CONTROLS	Yes Sector \times	Yes	Yes	Yes	Yes
FIXED EFFECTS		District, Sector	r × Year, and S	tate-Year Timi	5 Trends
Observations	36,985	18,712	35,818	18,712	35,818

Table E4: The Effects of Daily Temperature on Manufacturing Outcomes (Regulated)

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence (up to 1,800km) as modelled in Conley (1999) and serial correlation (1-year) as modelled in Newey and West (1987). District distances are computed from district centroids. Kernels are selected to provide the most conservative standard errors, looped over all distances between 10 and 2,000km and 1-7 years.

Table E5: The Effects of Daily Temperature on Manufacturing Outcomes (Regulated)

	Output	TFPR	TFPR	Items	
	Per Worker		(CES)	Produced	
Degree Days (10 days)	-0.00263*	-0.00214**	-0.00222**	-0.00140**	
$t_L=17, t_H=\infty$	(0.00159)	(0.000886)	(0.000919)	(0.000693)	
DD High \times Flexibility	0.00346	0.00206*	0.00212*	0.00157	
	(0.0244)	(0.0302)	(0.0142)	(0.0141)	
Degree Days (10 days)	-0.000404	-0.000242	-0.000244	-0.000107	
$t_L=0, t_H=17$	(0.000712)	(0.000388)	(0.000402)	(0.000296)	
DD Low \times Flexibility	0.000336	-0.0000634	-0.0000770	0.000251	
	(0.00107)	(0.000596)	(0.000615)	(0.000444)	
RAINFALL CONTROLS	Yes	Yes	Yes	Yes	
Fixed Effects	Sector \times District, Sector \times Year				
	and State-Year Time Trends				
Observations	36,985	33,445	33,464	36,985	

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence (up to 1,800km) as modelled in Conley (1999) and serial correlation (1-year) as modelled in Newey and West (1987). District distances are computed from district centroids. Kernels are selected to provide the most conservative standard errors, looped over all distances between 10 and 2,000km and 1-7 years.



Figure 4: The Effects of Daily Temperature on Manufacturing Outcomes (Regulated)



Figure 5: The Effects of Daily Temperature on Manufacturing Outcomes (Regulated)

E.2 Instrumental Variable Evidence

In this section I propose an Instrumental Variables strategy to support the basic identification strategy. As discussed, and demonstrated, in the main results, there are a number of channels through which temperature could affect manufacturing outcomes. Consequently, one cannot use temperature as an instrument for agricultural yields or employment to look at the effects of temperature on manufacturing through the labor reallocation channel, due to the clear violation of the exclusion restriction. However, if one believes that the labor regulation environment only moderates the effects of temperature through the labor reallocation channel then one can instrument the interaction between agricultural yields or employment and the labor regulation environment with the interaction of temperature and the labor regulation environment. This allows the level effect of temperature to continue serving its role in capturing the net effect of all remaining channels, while the interaction effect captures the labor regulaction channel.

Table E6 explores the first stages for this instrument that could be used on two endogenous regressors the share of workers who are agricultural laborers, and agricultural yields. In both cases increases in temperature in flexible labor market environments are associated with reductions in the share of employment and agricultural yields in flexible labor markets. However, the F-statistic is larger when the endogenous regressor is the interaction between labor market flexibility and agricultural yields. This may relate to the larger sample size due to the complete coverage of years.

	Ag. Worker Share × Flexible	\log Yields \times Flexible	
Daily Average Temperature	-16.225^{***}	-0.350^{***}	
× Flexible	(4.090)	(0.055)	
Fixed Effects	Sector × District, Sector × Yeah and State-Year Time Trends		
Observations	20,300	36,686	
Angrist-Pischke F-Stat	15.73	40.43	

Table E6: First Stage Estimates

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. The years of observation are 2003, 2004, 2005 and 2007 for the agricultural laborer instrument as these are the years in which data is available from the NSS. Data for agricultural yields is available for all years. Standard errors are clustered at the state level.

Focussing on the stronger first-stage relationship I estimate second-stage relationships for manufacturing firm outcomes, using the interaction of log Yields and the flexibility of the labor regulation environment as the endogenous regressor.³⁴ This gives an elasticity interpretation to the instrumented regressor and the outcome variables which are in log form. Consistent with the reduced form exercise in the main text I find that a reduction in agricultural yields in more flexible labor regulation markets is associated with an increase in output and contract workers, with a reduction in the wages of contract workers. In addition, the number of permanent workers is unchanged and the average wage of permanent workers increase. The estimated elasticity of substitution between contract and permanent workers using these estimates is 0.29, smaller than the reduced form measure (0.53), but still consistent with the premise that contract workers are complementary in the production process (Table E7). As in the reduced form results I also find that a reduction in agricultural yields in more flexible labor regulation markets is associated with increases in output per worker and the number of items produced, as well as noisily estimated increases in TFP (Table E8). Together these results provide suggestive evidence that the inflow of contract workers allow firms to move down the average cost curve, by allowing permanent workers to engage in more productive activities.

	log Output and Employment				
	Total Output	Items Produced	Employment (Contract)	Employment (Permanent)	
Daily Average Temperature (°C)	-0.0622 (0.0424)	-0.0363^{**} (0.0164)	-0.145^{***} (0.0398)	0.0174 (0.0327)	
log Ag. Yields \times Flexibility	-0.438^{**} (0.195)	-0.168^{**} (0.0802)	-0.806*** (0.256)	-0.176 (0.161)	
RAINFALL CONTROLS	Yes	Yes	Yes	Yes	
Fixed Effects	Sector \times Distr	ICT, SECTOR 2	× Year, and Stat	e-Year Time Trends	
OBSERVATIONS	36,686	36,686	18,421	35,521	

Table E7: Second Stage (log Yields \times Flexible): The Effects of Daily Temperature on Manufacturing Outcomes (Regulated)

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence (up to 1,800km) as modelled in Conley (1999) and serial correlation (1-year) as modelled in Newey and West (1987). District distances are computed from district centroids. Kernels are selected to provide the most conservative standard errors, looped over all distances between 10 and 2,000km and 1-7 years.

 $^{^{34}}$ Results are broadly robust to using the share of agricultural workers as the endogenous regressor presented in Tables E9 and E10.

	log Wages and Productivity					
	Avg. Day Wage (Contract)	Avg. Day Wage (Permanent)	Output Per Worker	TFPR	$\begin{array}{c} \mathrm{TFPR} \\ \mathrm{(CES)} \end{array}$	
DAILY AVERAGE	0.0202	-0.0578**	-0.0735**	-0.0737**	-0.0771**	
Temperature (°C)	(0.0164)	(0.0228)	(0.0375)	(0.0330)	(0.0370)	
log Ag. Yields	0.433^{**}	-0.236**	-0.366**	-0.277	-0.288	
\times Flexibility	(0.175)	(0.112)	(0.183)	(0.170)	(0.188)	
RAINFALL CONTROLS	Yes	Yes	Yes	Yes	Yes	
Fixed Effects	Sector \times District, Sector \times Year					
	and State-Year Time Trends					
OBSERVATIONS	18,421	35,521	36,686	33,137	33,163	

Table E8: Second Stage (log Yields \times Flexible): The Effects of Daily Temperature on Manufacturing Outcomes (Regulated)

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence (up to 1,800km) as modelled in Conley (1999) and serial correlation (1-year) as modelled in Newey and West (1987). District distances are computed from district centroids. Kernels are selected to provide the most conservative standard errors, looped over all distances between 10 and 2,000km and 1-7 years.

Table E9: Second Stage (Ag. Laborer Share \times Flexible): The Effects of Daily Temperature on Manufacturing Outcomes (Regulated)

	log Output and Employment				
	Total Output	Items Produced	Employment (Contract)	Employment (Permanent)	
Daily Average Temperature (°C) Ag. Laborer Share × Flexibility	-0.0619* (0.0365) -0.877*** (0.311)	-0.0548** (0.0258) -0.770*** (0.299)	-0.258*** (0.0784) -1.760* (0.900)	0.0101 (0.0804) -0.635 (0.846)	
RAINFALL CONTROLS FIXED EFFECTS	Yes Sector \times Distr	Yes ict, Sector >	Yes < Year, and Stat	Yes 'e-Year Time Trends	
Observations	20,300	20,300	10,190	19,615	

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence (up to 1,800km) as modelled in Conley (1999) and serial correlation (1-year) as modelled in Newey and West (1987). District distances are computed from district centroids. Kernels are selected to provide the most conservative standard errors, looped over all distances between 10 and 2,000km and 1-7 years.

		log Wages and Productivity				
	Avg. Day Wage (Contract)	Avg. Day Wage (Permanent)	Output Per Worker	TFPR	$\begin{array}{c} \mathrm{TFPR} \\ \mathrm{(CES)} \end{array}$	
Daily Average Temperature (°C)	0.0608^{***} (0.0184)	-0.0496^{**} (0.0211)	-0.0392 (0.0482)	-0.0577 (0.0560)	-0.0739 (0.0631)	
Ag. Laborer Share × Flexibility	0.847^{***} (0.319)	-0.485^{*} (0.259)	-0.377 (0.558)	-0.640 (0.473)	-0.750 (0.537)	
RAINFALL CONTROLS	Yes	Yes	Yes	Yes	Yes	
Fixed Effects	Sector \times District, Sector \times Year and State-Year Time Trends					
Observations	10,190	19,615	20,300	18,366	18,380	

Table E10: Second Stage (Ag. Laborer Share \times Flexible): The Effects of Daily Temperature on Manufacturing Outcomes (Regulated)

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Standard errors are adjusted to reflect spatial dependence (up to 1,800km) as modelled in Conley (1999) and serial correlation (1-year) as modelled in Newey and West (1987). District distances are computed from district centroids. Kernels are selected to provide the most conservative standard errors, looped over all distances between 10 and 2,000km and 1-7 years.

E.3 Alternative Definitions of the Labor Regulation Environment

In addition to the evidence in support of the identification strategy, I provide a series of robustness tests in support for the main results.

Table E11: Alternative Measures of the Labor Regulation Environment – (Neutral and Flexible Binary Variables)

	log Output and Employment			
	Total Output	Items Produced	Employment (Contract)	Employment (Permanent)
Daily Average Temperature (°C)	-0.0768^{*} (0.0449)	-0.0504^{***} (0.0162)	-0.135^{***} (0.0472)	0.00386 (0.0319)
Temperature × Neutral	0.0192 (0.0590)	0.0205 (0.0276)	0.00460 (0.0661)	-0.00509 (0.0468)
Temperature × Flexible	0.0888 (0.0647)	0.0380 (0.0264)	0.125 (0.0785)	$0.0566 \\ (0.0414)$
RAINFALL CONTROLS	Yes	Yes	Yes	Yes
Fixed Effects	Sector \times Distr	ICT, SECTOR >	× Year, and Stat	e-Year Time Trends
Observations	36,985	36,985	18,712	35,818

	log Wages and Productivity				
	Avg. Day Wage (Contract)	Avg. Day Wage (Permanent)	Output per Worker	TFPR	$\begin{array}{c} \mathrm{TFPR} \\ \mathrm{(CES)} \end{array}$
Daily Average Temperature (°C)	0.000705 (0.0263)	-0.0731^{***} (0.0169)	-0.0755^{*} (0.0415)	-0.0807^{***} (0.0250)	-0.0844^{***} (0.0260)
Temperature × Neutral	0.0345 (0.0337)	0.0247 (0.0234)	0.00638 (0.0525)	0.0231 (0.0265)	0.0297 (0.0266)
Temperature × Flexible	-0.0736^{*} (0.0383)	0.0493^{*} (0.0278)	0.0655 (0.0625)	0.0459 (0.0333)	0.0440 (0.0336)
RAINFALL CONTROLS	Yes	Yes	Yes	Yes	YES
Fixed Effects	Sector \times Dis	STRICT, SECTOR \times Y	EAR, AND STATE	-Year Time	Trends
OBSERVATIONS	18,712	35,818	36,985	33,440	33,464

Table E12: Alternative Measures of the Labor Regulation Environment – (Neutral and Flexible Binary Variables)

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. Panel A presents the results from regression the outcome variable on the level of temperature and monsoon rainfall. Panel B presents the differential effects of temperature across labor regulation environments. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

Table E13: Alternative Measures of the Labor Regulation Environment – (Flexible and Neutral Combined Binary Measure)

	log Output and Employment			
	Total Output	Items Produced	Employment (Contract)	Employment (Permanent)
Daily Average	-0.00281	-0.0131	-0.0380	0.0429^{*}
Temperature (°C)	(0.0351)	(0.0114)	(0.0336)	(0.0232)
Temperature	0.0487	0.0217^{*}	0.0705^{*}	0.0299
× Flexibility	(0.0324)	(0.0130)	(0.0383)	(0.0203)
RAINFALL CONTROLS	Yes	Yes	Yes	Yes
Fixed Effects	Sector \times Distr	ict, Sector >	× Year, and Stat	Te-Year Time Trends
OBSERVATIONS	36,985	36,985	18,712	35,818

		log Wages ani) Productivity		
	Avg. Day Wage (Contract)	Avg. Day Wage (Permanent)	Output per Worker	TFPR	$\begin{array}{c} \mathrm{TFPR} \\ \mathrm{(CES)} \end{array}$
Daily Average Temperature (°C)	-0.0373^{**} (0.0160)	-0.0256^{**} (0.0113)	-0.0246 (0.0310)	-0.0372^{**} (0.0179)	-0.0394^{**} (0.0184)
Temperature × Flexibility	-0.0371^{*} (0.0191)	0.0280^{**} (0.0141)	0.0354 (0.0311)	0.0260 (0.0167)	0.0254 (0.0169)
RAINFALL CONTROLS	Yes	Yes	Yes	YES	YES
Fixed Effects	Sector \times Dis ⁷	TRICT, SECTOR \times YI	EAR, AND STATE-	Year Time	Trends
Observations	18,712	35,818	36,985	33,440	33,464

Table E14: Alternative Measures of the Labor Regulation Environment – (Flexible and Neutral Combined Binary Measure)

E.4 Concerns Relating to the Endogenous Selection of Firms around the Regulatory Threshold

One concern relates to the endogenous selection of firms around the regulatory threshold. To mitigate these concerns I run the baseline specification dropping firms that have employment within 20% of the thresholds.

	log Output and Employment			
	Total Output	Items Produced	Employment (Contract)	Employment (Permanent)
Daily Average	-0.0933	-0.0458*	-0.180^{**}	0.0114
Temperature (°C)	(0.0572)	(0.0247)	(0.0717)	(0.0386)
Temperature	0.190^{**}	0.0486	0.254^{**}	0.0462
× Flexibility	(0.0931)	(0.0391)	(0.111)	(0.0572)
Rainfall Controls	Yes	Yes	Yes	Yes
Fixed Effects	Sector \times Distr	ict, Sector >	< Year, and Stat	re-Year Time Trends
Observations	32,475	32,475	16,624	31,631

Table E15: Baseline Specification - Dropping Firms with 20% of the Threshold

	log Wages and Productivity				
	Avg. Day Wage (Contract)	Avg. Day Wage (Permanent)	Output per Worker	TFPR	$\begin{array}{c} \mathrm{TFPR} \\ \mathrm{(CES)} \end{array}$
Daily Average Temperature (°C)	0.0307 (0.0355)	-0.0777^{***} (0.0237)	-0.0865^{*} (0.0523)	-0.0940^{***} (0.0293)	-0.0981^{***} (0.0310)
Temperature × Flexibility	-0.112^{**} (0.0526)	$\begin{array}{c} 0.0972^{***} \\ (0.0370) \end{array}$	0.141 (0.0869)	0.101^{**} (0.0431)	0.106^{**} (0.0445)
RAINFALL CONTROLS	Yes	Yes	Yes	Yes	Yes
Fixed Effects	Sector \times Dis	STRICT, SECTOR \times Y	EAR, AND STATE	-Year Time	Trends
OBSERVATIONS	16,624	31,631	32,475	29,489	29,510

Table E16: Baseline Specification - Dropping Firms with 20% of the Threshold

	1	og Output an	ND EMPLOYMENT	
	Total Output	Items Produced	Employment (Contract)	Employment (Permanent)
Panel A: Above 100 UP dropped				
DAILY AVERAGE	-0.169***	-0.0467*	-0.144**	-0.0225
Temperature (°C)	(0.0582)	(0.0240)	(0.0692)	(0.0429)
Temperature	0.215**	0.0555	0.172	0.102^{*}
× Flexibility	(0.0937)	(0.0365)	(0.108)	(0.0562)
Observations	35,102	35,102	17,716	33,980
Panel B: Above 120 UP dropped				
DAILY AVERAGE	-0.114*	-0.0373	-0.145**	0.00615
Temperature (°C)	(0.0656)	(0.0278)	(0.0708)	(0.0492)
Temperature	0.183*	0.0362	0.197^{*}	0.0478
× Flexibility	(0.104)	(0.0429)	(0.111)	(0.0693)
Observations	30,709	30,709	15,722	29,913
Panel C: Above 300				
DAILY AVERAGE	-0.162*	-0.0511	-0.253**	-0.0566
Temperature (°C)	(0.0862)	(0.0349)	(0.101)	(0.0623)
Temperature	0.308**	0.0790	0.412***	0.171^{*}
× Flexibility	(0.128)	(0.0532)	(0.151)	(0.0901)
Observations	14,264	14,264	8,144	14,048
Panel C: Above 360				
DAILY AVERAGE	-0.160*	-0.0510	-0.194*	-0.0491
Temperature (°C)	(0.0892)	(0.0374)	(0.111)	(0.0723)
Temperature	0.260**	0.0932^{*}	0.279^{*}	0.151
× Flexibility	(0.131)	(0.0558)	(0.167)	(0.106)
Observations	11,491	11,491	6,592	11,333

Table E17: Baseline Specification - Uniform Thresholds

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. UP dropped = Uttar Pradesh dropped. This is because the firm-size threshold for UP is 300. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

	log Wages and Productivity				
	Avg. Day Wage (Contract)	Avg. Day Wage (Permanent)	Output per Worker	TFPR	$\begin{array}{c} \mathrm{TFPR} \\ \mathrm{(CES)} \end{array}$
Panel A: Above 100 UP dropped					
Daily Average Temperature (°C)	0.0325 (0.0361)	-0.0786^{***} (0.0217)	-0.129^{**} (0.0520)	-0.0888^{***} (0.0323)	-0.0909^{***} (0.0336)
Temperature × Flexibility	-0.125^{**} (0.0537)	0.0800^{**} (0.0355)	0.140 (0.0854)	0.0802^{*} (0.0464)	0.0800^{*} (0.0471)
Observations	17,716	33,980	35,102	31,925	31,947
Panel B: Above 120 UP dropped					
Daily Average Temperature (°C)	0.0287 (0.0361)	-0.0789^{***} (0.0217)	-0.0826 (0.0520)	-0.0897^{***} (0.0323)	-0.0900^{***} (0.0336)
Temperature × Flexibility	-0.116^{**} (0.0555)	0.0941^{**} (0.0409)	0.117 (0.0930)	0.0849^{*} (0.0473)	0.0839^{*} (0.0486)
Observations	15,722	29,913	30,709	28,039	28,059
Panel C: Above 300					
Daily Average Temperature (°C)	0.0838^{*} (0.0473)	-0.0533^{*} (0.0323)	-0.0966 (0.0870)	-0.0938^{***} (0.0310)	-0.0798^{**} (0.0326)
Temperature × Flexibility	-0.184^{***} (0.0707)	0.0764 (0.0477)	0.170 (0.121)	0.116^{**} (0.0491)	0.0869^{*} (0.0517)
Observations	8,144	14,048	14,264	12,910	12,915
Panel D: Above 360					
Daily Average Temperature (°C) Temperature × Flexibility	0.0934* (0.0562) -0.178** (0.0833)	-0.0611^{*} (0.0362) 0.102^{*} (0.0531)	-0.144 (0.0885) 0.204^{*} (0.124)	-0.0710** (0.0348) 0.115** (0.0517)	-0.0630^{*} (0.0357) 0.0880 (0.0543)
Observations	6,592	11,333	11,491	10,423	10,428

Table E18: Baseline Specification - Uniform Thresholds

NOTES: Significance levels are indicated as * 0.10 ** 0.05 *** 0.01. UP dropped = Uttar Pradesh dropped. This is because the firm-size threshold for UP is 300. Standard errors are adjusted to reflect spatial dependence as modelled in Conley (1999) and serial correlation as modelled in Newey and West (1987). District distances are computed from district centroids. Results are robust to clustering standard errors at the State Level.

F Adjustment Costs and the Potential Gains from Reallocation

This appendix provides an upper bound estimate of the gains associated with removing any adjustment costs that impede the movement of casual workers into permanent manufacturing positions. As discussed wage gaps could also be explained by skill differences and so do not necessarily imply a misallocation of talent. Consequently, the lower bound associated with this exercise is zero.

F.1 Modelling the Potential Gains from Reallocation

To provide some insight into the potential gains from reallocation I introduce some economic structure to the data and explore quantitatively the impact of removing the distortion between casual manufacturing employment and permanent manufacturing employment – a naïve counterfactual in which it is assumed that all differences in wages are driven by misallocation.

To assess the potential gains from reallocation I compare a hypothetical output level in which labor is efficiently allocated across activities to observed output following a similar approach taken in other firm-level and sector-level studies of misallocation (Restuccia and Rogerson, 2008, Hsieh and Klenow, 2009, Vollrath, 2009, 2014, Gollin et al., 2014).

Assuming that each activity operates with a Cobb-Douglas production technology and maximises profits, the wage distortion τ can be identified from the first-order condition,

$$w_j = (1 - \alpha)\Lambda_j L_j^{-\alpha} \left[\frac{1}{\tau}\right]$$
(19)

where $\Lambda_j = p_j A_j$. The presence of adjustment costs, τ , will distort the amount of labor used in activity j compared to the level used in the absence of the distortion. As τ falls labor becomes relatively cheaper for activity j, and so the amount of labor that is utilised rises. In this context misallocation arises as the marginal revenue product of labor is not equalised across activities. To identify aggregate output two additional assumption are required. First, I assume that labor is perfectly substitutable across activities, i.e., there is no activity specific human capital. This implies that the total amount of labor in the economy is simply $L = \sum_j L_j$.³⁵ The second assumption is that prices are exogenously fixed, consistent with a small open economy in which all activities produce output that can

³⁵This assumption implies that the gains from reallocation are an upper bound of the upper bound; however, if one considers sector-specific human capital as a constraint to reallocation then relaxing this constraint is part of the problem.

be traded internationally.³⁶ Under these assumptions observed output in the economy with adjustment costs can be written,

$$Y = \left(\sum_{j} \Lambda_{j}^{1/\alpha} \left[\frac{1}{\tau}\right]^{1/\alpha}\right)^{\alpha} \left(\sum_{j} L_{j}\right)^{1-\alpha}$$

which follows from using equation 19 for each activity to solve simultaneously for the shares L_j/L , and then taking the sum of output across activities. In the presence of adjustment costs Y is below the output-maximising level. Consequently, one can estimate, given the structure imposed above, how much output would rise under the counterfactual in which these adjustment costs are removed. The counterfactual output after removing these adjustment costs is written,

$$Y^* = \left(\sum_j \Lambda_j^{1/\alpha}\right)^\alpha L_j^{1-\alpha}$$

With both observed output and counterfactual output levels, the gains from reallocation can be written as,

$$G = \frac{Y^*}{Y} = \frac{\left(\sum_j \Lambda_j^{1/\alpha}\right)^{\alpha}}{\left(\sum_j \frac{\Lambda_j^{1/\alpha}}{\tau^{1/\alpha}}\right)^{\alpha}}$$

providing a measure of the gains in aggregate productivity from eliminating the adjustment costs that impede the movement of labor across activities.

F.2 Estimating the Potential Gains from Reallocation

Given the model structure discussed above I estimate the gains from reallocation, \hat{G} , for each firm providing the average gains from reallocation, as well as the distribution of gains.

Under the naïve assumption that the only difference in wages across activities is driven by misallocation the average (observable) wage in the destination activity is,

$$\mathbb{E}[w_j] = \mathbb{E}[w_i]\tau,$$

In log-linear terms, the distortion can therefore be estimated as the log-difference in average wages across sectors,

 $^{^{36}}$ With endogenous prices the gains from reallocation would be smaller as an equivalent movement of workers out of casual activities raises the marginal revenue product of labor by more than if prices are held fixed.

$$\log \tau = \log \mathbb{E}[w_j] - \log \mathbb{E}[w_i]$$

Taking this to the data, I estimate the following moment for each firm,

$$\mathbb{E}[\tau] = \exp(\log \mathbb{E}[w_j] - \log \mathbb{E}[w_i])$$

In addition, I use estimates of the average permanent manufacturing wage, $\mathbb{E}[w_p]$, the average casual manufacturing wage, $\mathbb{E}[w_c]$, and the number of workers in each activity, L_i .

With these estimates, and an assigned value of α , I estimate output in each activity, Λ_i ,

$$\hat{\Lambda}_j = \frac{\hat{w}_j \hat{\tau}}{1 - \alpha} \hat{L}_j^{\alpha}$$

These values are then used to construct estimates for the observed level of output \hat{Y} , the counterfactual level of output \hat{Y}^* , and, with these estimates, the estimated gains from reallocation \hat{G} .

$$\hat{G} = \frac{\hat{Y}^*}{\hat{Y}} = \frac{\left(\sum_j \hat{\Lambda}_j^{1/\alpha}\right)^{\alpha}}{\left(\sum_j \frac{\hat{\Lambda}_j^{1/\alpha}}{\hat{\tau}^{1/\alpha}}\right)^{\alpha}}$$

F.3 Counterfactual Estimates

In considering the gains from reallocation I construct a counterfactual that removes the total wage gap across activities providing an upper bound on the size of adjustment costs. The results of this exercise are presented in table F1.

	(1)	(2)	(3)	(4)	(5)
Naïve Gains	1.196	1.131	1.089	1.062	1.043
Labor Share $((1 - \alpha))$	0.9	0.8	0.7	0.6	0.5

Table F1: The Average Output Gains from Reallocation

NOTES: These estimate provide an upper bound of the static gains from reallocation under the assumption that the total wage gap between casual manufacturing workers and permanent manufacturing workers are driven by adjustment costs. The lower bound estimate of the static gains from reallocation are therefore zero.

I estimate that the removal of adjustment costs τ_j would result in an 8.9% increase

in the manufacturing output of regulated firms hiring both casual and permanent workers ($\alpha = 0.3$), a non-trivial increase.

As emphasised, it is beyond the scope of this exercise to provide inferences about the relative contribution that adjustment costs may play in explaining the wage gap between casual and permanent manufacturing workers. Instead this exercise provides an upper bound on the gains from reallocation, under the assumption that the total wage gap is driven by adjustment costs. The lower bound is zero. Understanding the relative importance that adjustment costs play in impeding the movement of workers out of casual employment and into permanent positions remains an important area for future research.

CENTRE FOR ECONOMIC PERFORMANCE Recent Discussion Papers

1543	Andrew B. Bernard Esther Ann Bøler Swati Dhingra	Firm-to-Firm Connections in Colombian Imports
1542	Gerard H. Dericks Hans R.A. Koster	The Billion Pound Drop: The Blitz and Agglomeration Economics in London
1541	Andrew B. Bernard Andreas Moxnes	Networks and Trade
1540	Does Gentrification Displace Poor Households? An 'Identification-Via- Interaction' Approach	Sevrin Waights
1539	Zhuan Pei Jörn-Steffen Pischke Hannes Schwandt	Poorly Measured Confounders Are More Useful On the Left than On the Right
1538	Konstantin Büchel Stephan Kyburz	Fast Track to Growth? Railway Access, Population Growth and Local Displacement in 19 th Century Switzerland
1537	Stephen Gibbons Maria Sánchez-Vidal Olmo Silva	The Bedroom Tax
1536	Pawel Adrjan Brian Bell	Pension Shocks and Wages
1535	Tom Kirchmaier Stephen Machin Matteo Sandi Robert Witt	Prices, Policing and Policy: The Dynamics of Crime Booms and Busts

1534	Laurent Bouton Paola Conconi Francisco Pino Maurizio Zanardi	Guns, Environment and Abortion: How Single-Minded Voters Shape Politicians Decisions
1533	Giulia Giupponi Stephen Machin	Changing the Structure of Minimum Wages: Firm Adjustment and Wage Spillovers
1532	Swati Dhingra Rebecca Freeman Eleonora Mavroeidi	Beyond Tariff Reductions: What Extra Boost From Trade Agreement Provisions?
1531	Doruk Cengiz Arindrajit Dube Attila Lindner Ben Zipperer	The Effect of Minimum Wages on Low-Wage Jobs: Evidence from the United States Using a Bunching Estimator
1530	Stephen Gibbons Vincenzo Scrutinio Shqiponja Telhaj	Teacher Turnover: Does it Matter for Pupil Achievement?
1529	Ghazala Azmat Stefania Simion	Higher Education Funding Reforms: A Comprehensive Analysis of Educational and Labour Market Outcomes in England
1528	Renata Lemos Daniela Scur	All in the Family? CEO Choice and Firm Organization
1527	Stephen Machin Matteo Sandi	Autonomous Schools and Strategic Pupil Exclusion
1526	Stephan E. Maurer	Oil Discoveries and Education Spending in the Postbellum South

The Centre for Economic Performance Publications Unit Tel: +44 (0)20 7955 7673 Email <u>info@cep.lse.ac.uk</u> Website: <u>http://cep.lse.ac.uk</u> Twitter: @CEP_LSE